



Abdul Mannan Rauf ^{1,2,3,*}, Mohamed Abdel-Monem ¹, Thomas Geury ^{2,3} and Omar Hegazy ^{2,3}

- ¹ Research and Development (R&D) Unit, Powerdale (PWD), Witte Patersstraat 4, 1040 Brussel, Belgium
- ² MOBI Research Group, ETEC Department, Vrije Universiteit Brussel (VUB), Pleinlaan 2,
- 1050 Brussel, Belgium; omar.hegazy@vub.be (O.H.)
- ³ Flanders Make, Gaston Geenslaan 8, 3001 Heverlee, Belgium
- * Correspondence: abdul.mannan.rauf@vub.be

Abstract: Recently, multilevel converters (MLCs) have gained significant attention for stationary applications, including static compensators, industrial drives, and utility-grid interfaces for renewable energy sources. Compared to two-level voltage-source inverters (VSI) MLCs feature high-quality AC voltage with reduced harmonic content despite the lower switching frequency of the semiconductor devices. On the DC side, MLCs can integrate multiple isolated/non-isolated battery modules instead of a single battery pack. This helps to keep the system in service in case of a malfunction of one or more battery modules, as well as active balancing among the modules, a feature not possible with two-level VSI. In general, MLCs can be classified into two types: (i) two-port MLCs, which provide a single interface to connect with the battery pack, and (ii) multiport MLCs, which provide multiple interfaces to allow connection at the module or cell level. The classical topologies of both MLC types (e.g., neutral point clamped, flying capacitor, cascaded bridge) face limitations due to the high switch count. Consequently, many hybrid and reduced-switch topologies are reported in the literature. This paper presents a critical overview of both classical and recently reported MLC topologies and offers a better insight of MLC operation for grid-connected and standalone applications. In addition, the analysis thoroughly assesses various high-level control and modulation strategies while considering active balancing among the battery modules. Other salient features such as balancing speed during offtake/grid-injection mode and fault-ride-through capability are also incorporated. In conclusion, the key findings are summarized for a better understanding of the present and future integration of battery systems in stationary applications.

Keywords: multi-level inverters; stationary battery systems; hot swapping; grid; battery-energy storage; active balancing; modulation strategies

1. Introduction

In recent years, the deployment of battery-based energy-storage systems (BESSs) has been growing exponentially for stationary applications [1–5]. Being one of the core elements of an environmentally friendly and resource-efficient solution, BESSs have assumed the fundamental role of providing sustainable and high-standard energy sources. Grid-connected BESSs enhance grid reliability by stabilizing the power fluctuations caused by intermittent renewable-energy sources (RES), improving self-consumption and peak shaving for prosumers via a coordinated supply-and-demand time-shift [6,7]. In addition, BESSs provide ancillary services that include frequency-containment reserves (FCR), frequency-restoration reserves (FRR), imbalance mitigation, and replacement reserves [8–10]. In the off-grid mode, BESSs serve as the backbone of standalone power systems where there is no access to the utility grid.

Lithium-ion batteries (LiB) account for more than 80% of the installed capacity among different stationary BESS technologies due to their high energy and power density [11].



Citation: Rauf, A.M.; Abdel-Monem, M.; Geury, T.; Hegazy, O. A Review on Multilevel Converters for Efficient Integration of Battery Systems in Stationary Applications. *Energies* **2023**, *16*, 4133. https://doi.org/ 10.3390/en16104133

Academic Editor: Gianluca Brando

Received: 23 March 2023 Revised: 23 April 2023 Accepted: 26 April 2023 Published: 16 May 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). In a typical LIB BESS, multiple individual cells are first connected in series to make a module. The series connection adds up the individual cell voltages to the desired module-level open-circuit voltage (OCV). A typical module can have OCV in the range of 50–70 V [12]. The modules are then further connected in series and parallel combinations to form the complete battery pack. The parallel connections among the modules increase the usable capacity/energy of the pack, measured in kilowatt hours (kWh), whereas the number of parallel connections is a function of the use case [13–19]. For example, grid ancillary services (FCR, FRR) and prosumer use cases (self-consumption, arbitrage, peak shaving) require high-energy (HE) BESSs and therefore require a large number of parallel modules. On the contrary, decentralized generators require high-power (HP) BESSs to support the black-start capability. Unlike HE, the HP use case does not require many parallel connected modules [20].

The number of series connections impacts the pack OCV, which in turn depends on the number of stages in the PE interface. A single-stage PE interface consists of a single DC-AC inverter and therefore requires high DC-link voltage (>330 V for single-phase AC connection), as shown in Figure 1. This means that at least six modules (each 50–70 V) are required to be connected in series on the DC side. On the contrary, the two-stage PE interface does not apply any voltage restriction on the pack OCV, as there is an intermediate boostconversion stage before the DC-AC inverter. However, due to the higher cost associated with two stages, a single-stage PE interface is preferred [21–24]. Therefore, this review paper considers the single-stage PE interface as a reference.



Figure 1. Schematic overview of stationary BESSs, power electronics converter, grid-connected system, and off-grid load.

Another important component of a stationary BESS is a battery-management system (BMS). The BMS continuously monitors the critical pack parameters, which include minimum and maximum cell voltage/current, temperature, and state of charge (SOC). It ensures that the BESS operates inside the safe operating area (SOA). A BMS also provides balancing functionality among the individual cells. The balancing circuit can be passive (switched resistor bank), which results in higher power dissipation, or active (switched capacitors and inductors), which results in high implementation costs [25–28].

Conventionally, a two-level voltage-source inverter (VSI) is used for a single-stage PE interface, as shown in Figure 1. The DC port of the two-level VSI is connected to the battery-pack terminals, whereas the AC port is connected to the grid via the filter inductor or directly to the standalone AC load. Due to fewer switches, the two-level VSI is simple in structure but faces multiple downsides.

Firstly, it faces a potential capacity reduction on the BESS side. As it requires a large number of series connections in the battery pack, the individual cells accumulate differences in SOC and reach the cut-off voltage earlier than other cells. This happens mainly due

to non-homogeneity and different internal impedances/self-discharge rates of individual cells. To avoid damages due to overheating or a deep state of discharge, the BMS must disconnect the whole battery pack.

Secondly, the two-level VSI operates at a higher switching frequency, which exposes the semiconductor switches and interfacing filter to high dv/dt. In the standalone mode where there is no filter inductor, the high dv/dt and small rise times create undesired effects, which include excessive voltage stress on windings, leakage currents, and increased EMI. Furthermore, if long cables are used to connect the drive, the rapid rise times can potentially cause the transmission-line effect, which can double the voltage applied to the drive terminals, as explained in [29].

Considering the downsides of two-level VSI, this review paper focuses on multilevel converters (MLCs) as a reliable BESS interface for on-grid and off-grid applications. MLCs were first introduced in 1975 by Baker and Lawrence as an improved alternative to a two-level VSI in high-power applications [30]. The output AC voltage of MLCs is a synthesized multi-step (greater than two) signal due to the switching of multiple semiconductor switches. The number of levels in the AC voltage is directly proportional to the number of switches. Compared to two-level VSIs, MLCs offer a higher degree of flexibility, as they can be connected at the pack, module, or individual cell level. They offer high-quality AC voltage, lower switching losses, lower voltage stress on semiconductor switches, and reduced filter size for grid connection [30–33]. Some drawbacks include the large number of semiconductor switches and associated gate-driver circuitry, which can lead to a high implementation cost [33].

An important feature of a BESS-fed MLC is the potential to perform active balancing among the battery modules, a feature not possible with a two-level VSI. Active balancing is an added value of the MLC system that eliminates the requirement of a separate BMS. It is an important performance parameter to consider in selecting an efficient MLC topology for stationary BESSs. However, in the MLC literature, no review articles are available that cover this aspect for stationary BESSs [34,35]. Most review articles dedicated to RES-fed MLCs are limited only to PV systems [34–38]. Other review articles have focused on grid-connected BESSs but do not consider the battery-balancing requirements [30–33].

The main contribution of this paper is to present a critical overview of the classical, hybrid, and reduced-switch MLC topologies for stationary BESSs. The topologies are analyzed based on different key performance indicators (KPIs), which include (i) switch/diode count, (ii) passive-component count, (iii) AC-mode support, (iv) active balancing, (v) modularity, (vi) dual-mode support (grid-connected to off-grid transition), (vii) switch utilization, (viii) fault ride through capability, and (ix) DC-microgrid support. In addition, an overview of the latest control and modulation techniques is also presented. The analysis takes into account the dual role of modulation technique for BESS-fed MLCs, which ensures that the operating principle of active balancing does not violate while also maintaining a good-quality AC voltage.

The paper is organized as follows. Section 2 presents and classifies various potential MLC topologies for stationary BESS. It also presents the active-balancing principle and the role of multiport MLCs for stationary BESSs. Section 3 presents the topological comparison based on the aforementioned nine KPIs. Section 4 explains different modulation strategies, and Section 5 presents the higher-level control strategies. A simulation study is conducted in Section 6 to compare various active-balancing modulation techniques. Finally, Section 7 concludes the paper by summarizing the key findings and directions for future research.

2. Topologies of MLC Systems for Stationary BESSs

For grid-connected systems, the main requirement for an independent stationary BESS is to balance the offtake and generation for a specific group of demand (loading) points.

This is normally done by equalizing the fluctuations in the electricity generation and demand by time-shifting the load requirement. For an off-grid system, maintaining an uninterrupted power supply (UPS) is the main requirement [39,40]. Figure 2 reflects the classification of MLCs for a stationary BESS. Depending upon the type, isolation, and arrangement of the battery pack/modules, MLC systems can be broadly classified into two categories: (i) two-port and (ii) multiport MLCs.



Figure 2. Categories of MLC configurations for stationary BESSs.

The two-port MLC provides a single interface and is connected to the whole battery pack, whereas multiport MLC provides multiple interfaces to connect several modules or individual cells. The two-port MLC does not provide balancing functionality but significantly decreases the THD in the AC voltage compared to a two-level VSI. It also allows a smaller interfacing filter for grid connections. The sub-section below briefly covers the underlying topologies for the two-port MLC system.

2.1. MLC with Two-Port Interface

Figure 3b–g reflect the potential configurations for a two-port MLC. Figure 3a depicts the grid-connected and off-grid AC interface to evaluate both modes of operation for each configuration in stationary applications.

2.1.1. Neutral-Point Diode Clamped (NPC)

In the NPC topology, the input DC voltage is divided into an even number of ACvoltage levels. For x levels in the AC voltage V_{Ac} , the NPC MLC consists of 2(x - 2)clamping diodes, (x - 1) DC-link capacitors, and 2(x - 1) semiconductor switches [41,42]. Figure 3b reflects the NPC MLC topology for five-level V_{Ac} . It has the advantage of bipolar DC bus operation but lacks modularity and balancing of the DC-link capacitors. In addition, it faces limitations due to uneven power losses and unequal blocking voltage across six diodes.

Another variant of the NPC topology is the active-NPC topology reported by ABB in [43]. It adds up the phase capacitor in a three-level NPC configuration to produce two additional voltage levels. Active-NPC topology finds widespread applications for industrial drives.

2.1.2. Flying-Capacitor MLC

In this topology, the MLC consists of 2(x - 2) clamping diodes, (x - 1) DC-link capacitors, and 2(x - 1) power electronics switches for single-phase x-level AC voltage [44–46]. Figure 2c depicts the single-phase flying-capacitor MLC topology for five-level V_{Ac} . The single-phase



flying-capacitor topology consists of four identical DC-link capacitors $C_1 - C_4$ of the same voltage rating for splitting the DC bus into four equal voltages.

(b)





4 Cascaded half bridges HB1-HB4

(f)



Figure 3. (a) Grid-connected and standalone load for MLC. Multiple configurations of the two-port MLC system fed by single/non-isolated BESSs. These include (b) NPC topology, (c) flying-capacitor topology, (d) cascaded-T configuration, (e) hybrid HB-FC topology, (f) hybrid SC-HB topology, (g) hybrid MLC (H8) topology, and (h) hybrid FC-ANPC topology.

This topology is similar to NPC topology, as is clear from the comparison of Figure 3b,c, except that there are no clamping diodes.

The flying-capacitor MLC features higher modularity compared to the NPC topology and offers better voltage-forming capability, higher frequency operation, and active and reactive power management. These features make the flying-capacitor MLC a better candidate for a standalone power supply. However, it faces high switching losses while transferring the active power from the DC to the AC side and requires a significant number of capacitors, which reduces its economic viability compared to the NPC MLC [45].

2.1.3. Cascaded-T MLC

As reported in [45,47] a cascaded-T MLC is a reduced-switch MLC topology where a single-phase grid-connected or standalone x-level output consists of (x + 1) semiconductor switches, (x + 3) clamping diodes, and (x + 3) DC-link capacitors. Figure 3d depicts the cascaded-T MLC topology for nine-level output voltage.

2.1.4. Hybrid MLC

This combines two traditional MLC topologies to achieve higher voltage levels for the same number of switches. Ref. [48] reported the HB-FC topology, where four cascaded half bridges are combined with a seven-level flying capacitor MLC to achieve 15 voltage levels, as shown in Figure 3e.

In [49], the authors reported another hybrid MLC topology, where a combination of four cascaded half bridges and a three-level switched-capacitor (SC) converter generates nine levels, as reflected in Figure 3f. Figure 3g depicts another hybrid MLC (H8) topology, as reported in [48], where fewer switches achieve higher voltage levels than either of the parent topologies.

In [50], a combination of an active-NPC (ANPC) and a flying-capacitor MLC is reported, as shown in Figure 3h.

2.2. MLC with Multiport Interface

Multiport MLCs feature higher flexibility compared to two-port configurations, as they also allow connection at the module or individual cell level. Importantly, multiport MLCs can realize the terminal voltage or SOC balance between the battery cells or modules, and therefore do not suffer from capacity reduction due to the series connection among the cells [51–53].

If the MLC is connected at the cell level, the overall switch count becomes too high to be economically viable. As explained in Section 1, a 3.6 V LiB cell implies that more than 100 series-connected cells are needed per phase, which means that at least 400 semiconductor switches are required for single-phase grid connection. In the case of module-level implementation, fewer switches (less than 30) are needed, as each module is composed of multiple series-connected cells. The choice of cells per module is a non-technical constraint and is more based on the commercial availability of battery modules [54,55].

In this paper, 16 series-connected cells are considered in each module. Compared to a cell-level MLC, a module-level MLC faces a tradeoff, as the faulty cell cannot be isolated individually. However, it can be addressed by controlling the charging/discharging current of the module carrying the faulty cell. Moreover, each module still has a cell-level monitoring system (secondary BMS) to monitor the voltage/temperature of individual cells; however, it would not disconnect the faulty cell, as this feature is already supported by the module-level MLC.

2.2.1. Active Balancing with a Multiport MLC

The underlying building block for multiport MLCs is either a full-bridge (FB) or halfbridge (HB) structure that is interfaced with an isolated battery module [56,57]. Figure 4a,b reflect a single battery module with an HB and FB interface, respectively.



Figure 4. Operating modes of full-bridge and half-bridge interfaces with a single battery module: (a) generic full-bridge structure, (b) module-bypass state during discharging mode for full bridge, (c) generic half-bridge structure, and (d) module-bypass state during discharging state for half bridge.

Tables 1 and 2 show the operational modes of FB and HB interface as a function of various switching states, respectively [58].

State		Output	FB Switch State				Battery-Module	Battery-Module	Output
Mode	Controller <i>m_n</i>	Voltage (V _o)	S ₁₁	S ₁₂	S ₁₁	S ₂₂	State	Current (<i>i</i> _b)	Current (i _o)
1	1	$+V_{\rm B}-2V_{\rm d}$	1	0	0	1	Charging	Negative	Negative
2	1	$+V_{\rm B}-2V_{\rm d}$	-	0	0	-	Discharging	Positive	Positive
3	0	0	1	1	0	0	Bypassed	0	Positive
4	0	0	0	0	1	1	Bypassed	0	Negative
5	-1	$-V_{\rm B} + 2V_{\rm d}$	0	1	1	0	Charging	Positive	Positive
6	-1	$-V_{\rm B} + 2V_{\rm d}$	0	1	1	0	Discharging	Negative	Negative

Table 1. Operational modes of full-bridge interface as a function of switch states.

Table 2. Operational modes of half-bridge interface as a function of switch states.

	State	Output	HB Switch State		Battery Module	Battery Module	Output
Mode	Controller <i>m_n</i>	Voltage (V _o)	<i>S</i> ₁	<i>S</i> ₂	State	Current (<i>i</i> _b)	Current (i ₀)
1	1	$+V_{\rm B}-V_{\rm d}$	1	0	Charging	Negative	Negative
2	1	$+V_{\rm B}-V_{\rm d}$	-	Ũ	Discharging	Positive	Positive
3	0	0	0	1	Bypassed	0	Positive
4	0	0	0	1	Bypassed	0	Negative
5	-1				NTA		
6	-1	_			INA		

Comparing Tables 1 and 2, it can be seen that the FB interface provides three output voltages, $+V_B$, 0, and $-V_B$, through different combinations of S_{11} , S_{12} , S_{21} , and S_{22} . On the other hand, HB can provide two output voltages, $+V_B$ and 0 V, through different combinations of S_1 and S_2 .

The state transition is governed by the control parameter m_n , which can take values between -1 and 1. When $m_n = 0$, the output voltage $V_0 = 0$ and the battery module is bypassed from the MLC system, representing a no-charge/discharge state. The module current $i_b = 0$ but the output current (i_o) continues, which can be positive or negative.

Figure 4b,c show negative and positive output-current paths, respectively, for the FB and HB interface during the module-bypass state when $m_n = 0$ and $V_0 = 0$. When $m_n = 1$ or -1, V_0 and the real power (P_0) of the FB and HB interface can be expressed as follows:

$$V_{o-FB} = V_b - 2V_d \tag{1}$$

$$P_{o-FB} = V_b \cdot i_{AC} \cdot m_n \quad \forall -1 \le m_n \le 1$$
⁽²⁾

$$i_{o-FB} = \left\{ \begin{array}{ll} i_b & \text{if} \quad m_n = +1 \text{ or } -1 \forall V_0 \neq 0 \\ > 0 & \text{if} \quad m_n = 0 \forall V_0 = 0, i_b = 0 \\ < 0 & \text{if} \quad m_n = 0 \forall V_0 = 0, i_b = 0 \end{array} \right\}$$
(3)

$$V_{o-HB} = V_b - V_d \tag{4}$$

$$P_{o-HB} = V_b \cdot i_o \cdot m_n \qquad \forall \ 0 \le m_n \le 1 \tag{5}$$

$$i_{o-HB} = \left\{ \begin{array}{ccc} i_b & \text{if } m_n = +1 \ \forall \ V_0 \neq 0 \\ > 0 & \text{if } m_n = 0 \ \forall \ V_0 = 0, i_b = 0 \\ < 0 & \text{if } m_n = 0 \ \forall \ V_0 = 0, i_b = 0 \end{array} \right\}$$
(6)

where V_d is the voltage drop across one switch during the ON state. V_d can be positive or negative based on the direction of i_o . Comparing Equations (1) and (4), it can be seen that FB has a voltage drop two times higher than that of HB. In addition, except for the bypass mode, the output current in both topologies is equal to i_b .

For $0 \le m_n \le 1$, the charged or discharged energy of each battery module depends on the period during which the module is connected to the circuit.

Comparing Tables 1 and 2, FB allows an extra mode (Mode 6), where $V_0 = -V_B + 2V_D$. This state allows the generation of negative voltage levels in the output. The HB interface does not have Mode-6; however, it does not impact the balancing potential, as cascaded HBs can operate at a higher frequency than FB.

The choice of m_n can be used for the voltage or SOC equalization among the modules. Since each module is controlled by a dedicated m_n , the switch-in and switch-out time of each module can be controlled independently. The module with higher voltage or SOC can be discharged for longer duration (more $-V_B$ intervals) or charged for a shorter duration (more 0 intervals). m_n can also improve the energy-utilization ratio while avoiding the overcharging or deep discharging of individual modules.

2.3. Subtypes of MLC with Multiport Interface

The subsections below highlight several potential multiport MLC topologies for connections at the module level.

2.3.1. Cascaded H-Bridge (CHB) MLC

CHB topology consists of multiple cascaded FBs and generates AC voltage by switching between several small DC voltage levels using low-frequency switching devices [59–61]. Due to the modular structure, it holds an advantage in fault-tolerant operation over FC and NPC topologies, as there are no clamping diodes, bulky inductors, or capacitors in a CHB configuration. Typically, CHB consists of several FBs (h) connected in series. Each FB is fed by an isolated battery module. The output voltage V_{AC} and number of levels N_{levels} can be expressed as follows [61]:

$$V_{AC-max} = V_{b-max} \times h \tag{7}$$

$$N_{level} = 2h + 1 \tag{8}$$

where V_{AC-max} and V_{b-max} are the maximum values of output AC voltage and module DC-link voltage, respectively.

1

Figure 5 reflects a 13-level CHB MLC topology where six FBs are cascaded together. Each FB is connected to an individual battery module. The AC voltage has six levels (V_{b1} to $V_{b1} + V_{b2} \dots + V_{b6}$) in both positive and negative half cycles.



Figure 5. Schematic diagram of a multiport 13-level CHB system comprising six isolated battery modules.

CHB topology can operate at different modulation indexes for the FBs. This feature allows the CHB to perform active balancing, as it can maintain one module DC-link voltage at values independent of other modules. For a multiphase system, it can perform compensation of unbalanced phase currents.

Based on the voltage of the individual battery modules, CHB topology can be subclassified into symmetrical (SCHB) [62,63] and asymmetrical (ACHB) configurations [64,65].

In S-CHB, all the DC voltages are equal. In terms of components, SCHB needs $2(N_{levels} - 1)$ switches and $(N_{levels} - 1)$ /modules to generate N_{levels} in the AC voltage.

In the A-CHB, the DC voltages are kept different based on a geometric progression factor (GPF). The GPF commonly has a value of two or higher to generate more steps in the AC voltage without increasing the number of FBs. Accordingly, N_{levels} can be expressed as follows:

$$N_{level} = 2^{h+1} - 1 \quad \text{if} \quad V_k = 2^{k-1} V_b \ \forall \quad k \in [1, 2, 3, \dots, h]$$
(9)

$$N_{level} = 3^h \quad \text{if} \quad V_k = 3^{j-1} V_b \ \forall \qquad \mathbf{k} \in [1, 2, \dots, h] \tag{10}$$

The maximum AC voltage of *h* FBs can be expressed as:

$$V_{AC-max} = (2^{h} - 1) V_{b-max} \quad \text{if} \quad V_{k} = 2^{k-1} V_{b} \forall \quad \mathbf{k} \in [1, 2, \dots, h] \\ V_{AC-max} = (\frac{3^{h} - 1}{2}) V_{b-max} \quad \text{if} \quad V_{k} = 3^{k-1} V_{b} \forall \quad \mathbf{k} \in [1, 2, \dots, h] \end{cases}$$
(11)

Comparing Equations (6)–(11), it becomes clear that ACHB can generate more voltage levels and a higher amplitude of the AC voltage than SCHB with the same number of FBs. In the reference case, SCHB needs six modules to generate 13 levels with a peak AC voltage of 365 V, whereas AHCB can generate the same voltage with only three modules. However, there is a tradeoff, as ACHB is less modular and fault tolerant compared to S-CHB.

In the case of binary DC-link module voltages, ACHB yields only the additive voltage levels, whereas in trinary sequence selection of DC voltages, both additive and subtractive voltage combinations can be obtained. Another disadvantage is the requirement of modules with higher OCV, which means a large number of series-connected cells per module, thus increasing the balancing problems of the cells within the module.

2.3.2. Reduced-Switch MLC

Recently, many reduced-switch MLC configurations have been reported [35,36,38,66–76] to improve the number of levels and voltage harmonic profile without changing the component rating. Most of the configurations focus on MLCs interfaced with solar PV, where the reduction of switches allows only unidirectional power flow (PV panels to the AC side) and therefore do not support battery-module integration [35,36,38]. Likewise, many topologies do not allow the individual modules to be bypassed or re-ordered in the voltage-building process and therefore do not support the module-balancing feature [35]. The subsections below cover only reduced-switch topologies that are bidirectional and can perform the primary balancing among the isolated battery modules.

Reduced Cascaded Half-Bridge MLC

The reduced cascaded half-bridge (RCHB) MLC consists of one FB and multiple HBs connected in series. It is reported in [77] as an improvement over CHB-MLC because it requires a lower number of semiconductor switches to generate the same levels in AC voltage.

Figure 6 reflects the single-phase configuration of a 13-level RCHB-MLC where each HB is connected to an isolated battery module. The output of these cascaded HBs is connected to an intermediate DC bus. The DC bus exhibits a staircase-shaped periodic-voltage waveform at twice the frequency of the AC voltage. The DC bus is connected to an FB that simply inverts the DC voltage to generate the AC waveform. The FB uses low-frequency semiconductor switches that operate at the base frequency (50/60 Hz) of the fundamental AC voltage.

In RHCB topology, different battery modules can be involved or bypassed in the voltage-building process, as each HB can be controlled independently, thus achieving the primary module-balancing feature. Based on the variation in OCV of different modules (all the same, binary relation or other), the total number of levels ($N_{level-HCHB}$) and maximum valued of the AC voltage ($V_{AC-HCHB-max}$) can be expressed as follows:

$$N_{level-HCHB} = \left\{ \begin{array}{ccc} n+1 & \text{if } & V_{bk} = V_{DC} \ \forall & k \in [1,2,3,\dots,n] \\ 2^n & \text{if } & V_{bk} = 2^{k-1}V_{DC} \ \forall & k \in [1,2,3,\dots,m] \\ 2n & \text{if } & V_{bk} = 2V_{DC} \ \forall & k \in [1,2,3,\dots,n] \end{array} \right\}$$
(12)

$$V_{AC-HCHB-max} = \left\{ \begin{array}{ccc} n \times V_{DC} & \text{if } V_{bk} = V_{DC} \ \forall \ \mathbf{k} \in [1, 2, 3, \dots, n] \\ (2^n - 1)V_{DC} & \text{if } V_{bk} = 2^{k-1}V_{DC} \ \forall \ \mathbf{k} \in [1, 2, 3, \dots, m] \\ (2n - 1)V_{DC} & \text{if } V_{bk} = 2V_{DC} \ \forall \ \mathbf{k} \in [1, 2, 3, \dots, n] \end{array} \right\}$$
(13)

where *n* is the number of cascaded HBs per phase.



Figure 6. Schematic diagram of multiport 13-level RCHB topology comprising six isolated battery modules.

Comparing Figures 5 and 6, RHCB topology requires two switches per module, whereas CHB topology needs four switches per module to produce the same number of levels in the AC voltage. The total number of switches for both topologies can be expressed as follows [77]:

$$N_{sw-CHB} = 4 \times Modules N_{sw-RHCB} = 2 \times Modules + 4$$
(14)

This means that for a six-module MLC, $N_{sw-CHB} = 24$ and $N_{sw-HCHB} = 16$. The difference becomes larger as the module count increases.

The output of cascaded HBs is the multistep half-sinusoidal-shape DC voltage (V_{DC-h}) with a frequency two times higher than that of the AC voltage, as shown in Figure 6. The final FB converts V_{DC-h} to sinusoidal AC voltage by simply reversing the polarity of alternate cycles.

Despite the switch-count reduction, the RHCB topology faces a critical limitation in grid-connected mode that is not reported in [72,77] or in the available literature on the cascaded HB MLC [69,74]. The limitation is explained as follows.

RHCB uses the front-end FB for polarity generation, which means FB is essentially uncontrolled from the current-control perspective. The switches in FB are synchronously turned on and off at the zero crossings of the AC voltage and therefore lack the ability to inject current in the grid-connected mode.

The synchronous switching allows FB to operate in voltage-control mode, which is possible for standalone/off-grid applications, as explained in [72,77]. However, in the case of grid connection, the RCHB cannot inject power, as active and reactive currents of the

interfacing inductor (L_g) cannot be controlled with synchronous (50 Hz) switching of the semiconductor switches in FB. However, in the offtake mode it draws uncontrolled current, as the FB simply behaves like a diode rectifier due to synchronous switching.

Another option to perform indirect current control via cascaded HBs is also possible but adds high THD in the DC bus voltage V_{DC-h} since each HB operates as an independent DC-DC (synchronous buck) converter. The resulting voltage level in the intermediate DC bus needs a certain settling time before the next level can be added. For a grid-following operation, all six levels must be completed in half-cycle time (≤ 10 ms). This means that the settling time should be less than 1.6 ms per level for six HBs. Such a requirement creates an underdamped response with high overshoot in each level of V_{DC-h} . As a result, the THD in the AC voltage exceeds the allowable limit by the grid codes.

Modified Cascaded MLC

A modified cascaded MLC (MCMLC) is reported in [66] that allows a wide operating range, smaller THD, and higher efficiency for grid-connected mode. Figure 7a reflects the schematic of MCMLC. In this configuration, there is an auxiliary bidirectional switch S_{aux-1} between two cascaded FBs. The auxiliary switch allows the modules to be connected in both series and parallel configuration. When S_{aux-1} is turned off/bypassed, MCMLC functions as a normal CHB (as in Figure 5) connecting the modules via FBs (cascaded mode). When S_{aux-1} is turned on, the two modules are connected in parallel (HB mode), sharing the same negative DC bus. The parallel connection allows the higher current injection into the grid using the same switches.



Figure 7. Schematic diagram of reduced-switch MLC for (**a**) modified cascaded CHB and (**b**) packed U-cell configuration.

Despite a wide operational range, the MCMLC configuration has a few downsides. Firstly, the transformation between cascaded and HB mode requires the interruption of the inductor current, which can lead to voltage spikes across the semiconductor switches. To avoid these spikes, the mode transition must be done at zero-crossing points of the reference-modulating signal (zero voltage, zero current), which is only possible if the voltage and current are in-phase. This means that the MCMLC cannot control the reactive power and must operate at a unity power factor.

Secondly, the zero-voltage–zero-current transition cannot be achieved in a standalone system for stationary applications, as the loads are mostly inductive/nonlinear. As a consequence, the MCMLC can only function either in CHB or HB mode for off-grid applications.

Packed U-Cell MLC

The packed U-cell (PUC) topology is proposed in [75,76] as a reduced-component alternative to the CHB, NPC, and FC configurations. It consists of six semiconductor switches and two isolated battery modules, as shown in Figure 7b. A PUC MLC generates seven-level AC voltage using six switches. The upper switches, S_{11} and S_{12} , operate at the fundamental frequency of the AC voltage. The lower four switches operate at a relatively higher frequency and lower harmonics, thus reducing the switching losses of the PUC MLC. The left set of switches (S_{11} , S_{12} , S_{61}) operates in a manner complementary to the right set of switches (S_{12} , S_{22} , S_{62}).

The PUC topology is capable of operating in both grid-connected and standalone applications [51]. It can provide an uninterrupted power supply (UPS) for a wide range of changes in the load and module voltage. However, a major limitation of PUC topology is the voltage dependency of both modules. The OCV for module–2 (V_{b2}) needs to be regulated at one-third of the magnitude of V_{b1} . Such restrictions reduce the modularity of a PUC MLC.

2.3.3. Modular MLC

The generic implementation of a single-phase modular MLC (MMLC) comprises an AC port, a DC port, and the converting structure, consisting of two identical arms that are connected via two uncoupled inductors. Each arm consists of N submodules (SMs). Each SM constitutes the fundamental building block of the MMLC. The number of SMs per arm indicates the levels in the output AC voltage. Figure 8a reflects the single-phase MMLC schematic for the reference case with six battery modules. As shown in Figure 8a, there are six SMs. Each SM is interfaced with a single-battery module.

In the literature, numerous configurations are reported for SMs [4,78–82]. Two of the most well-known topologies that are widely used in MMLCs are HB and FB configurations [4]. As shown in Figure 8c, HB is relatively simple and cost effective but faces many limitations. It cannot stop the DC fault current in the converting structure. Due to its four switches, it can only produce two voltage levels: V_b with S_1 ON, S_2 OFF and 0 with S_1 OFF, S_2 ON. This means that the SMs can only be bypassed in the case of a short circuit but cannot stop the DC fault current. The FB configuration shown in Figure 8b can counter this issue, as it generates three levels: V_b , 0, and $-V_b$. Due to an extra level, $-V_b$, FB can offer high impedance and effectively block the DC fault current.

Besides HB and FB topologies, SMs can also use many configurations from a family of two-port or multiport MLC topologies, as reflected in Figure 2. Consequently, SMs will feature the same pros and cons associated with these topologies, as reported in [78].

Recently, [4] reported a compact, low-volume SM topology that uses stacked switched capacitors (SSC) acting as an energy-storage buffer. Figure 8d reflects the SSC configuration, where C_1 and C_2 make the energy buffer in conjunction with the battery module V_b . Compared to the HB SM configuration, it can decrease the total volume of SM by more than 40%.

To increase the levels in the SM voltage, cross-connected (CS) SM topology is reported in [83]. Figure 8e reflects the CSSM configuration, where two switches are used to connect back-to-back half-bridge SMs. This configuration has the advantage, as it tolerates the DC fault current by turning off the crossed switches.



Figure 8. (a) Schematic diagram of a modular MLC in a single-phase system. Various configuration of SM, including (b) FB, (c) HB, (d) stacked switched capacitors, and (e) cross-connected SM configuration.

Similarly, switched-capacitor (SC) SMs and composite three-level SMs (CSMs) are reported in [84,85], respectively. The SCSM can ride through a DC-link short circuit by turning off all the active switches, whereas the CSM can address the DC fault current together with unbalanced charging of the switched capacitors.

2.3.4. Hybrid MLC

Researchers have proposed the concept of hybrid MLC to combine two or more MLC topologies to achieve higher voltage levels. However, not all the hybrid topologies in the literature [86,87] are relevant for battery-module integration due to the constraint of bidirectional power flow and module-balancing requirements, as explained in Section 2.2.1. This includes the hybrid topology, where the NPC and FC are merged through a coupled inductor [87] and the nine-level T-Type NPC and FCH-bridge configurations reported in [86].

An interesting asymmetric cascaded hybrid MLC configuration (ACHHB) is introduced in [88], where several HBs are cascaded with one FB, as shown in Figure 9a. The DC-link voltage of HBs can be kept equal or follow a binary sequence, keeping the DC source of FB constant. ACHHB can generate up to 13 levels with five HBs and one FB unit, compared to the six FB units required for CHB topology, and considerably improves THD in the output voltage. If the DC voltage of HB units vary in binary or trinary fashion, the number of levels in AC voltage can be much higher [88].



Figure 9. Schematic diagram of a hybrid MLC for (**a**) asymmetric cascaded configuration and (**b**) ANPC-CHB configuration.

Figure 9b depicts a hybrid ANPC structure with CHB topology. The purpose of introducing CHB units is to increase the number of levels without any variation in overall power. The CHB units act as active power filters to enhance the quality of the AC voltage. Similar to [88], another modular half- and full-bridge hybrid (MHFH) topology is reported in [89], which offers extra levels and can be extended in a cascaded manner. However, due to the presence of two unidirectional switches, this topology cannot charge all the modules simultaneously.

3. Topological Comparison

Table 3 presents a summary of the reviewed two-port and multiport MLC topologies for stationary BESSs. The type of topology and the reporting publication are included for better understanding. For multiport MLCs, the base case of six modules is considered as a reference for each topology. Based on the active-balancing principle in Section 2.2.1, seven state-of-the-art multiport MLC topologies that can perform active balancing among the battery modules were shortlisted. These topologies include (i) symmetric CHB [62,63], (ii) asymmetric CHB [64,65,67,88], (iii) RCHB [35,36,38,66–76], (iv) MCMLC [4,79,85], (v) HB/FB MLC [81], and (vi) ACHHB [90]. The passive-element count in Table 3 also includes the filter inductor for a grid interface.

Table 3. Comparison of core technical features among various MLC topologies.

Topology	Class	Ref	Structure	Switch (Diode) Count	Passive Element (Count)	AC Mode	Active Balancing	AC Levels	Merits/Demerits
NPC	MLC with	[41-43]	Traditional	8(6)	C(4) L(1)	*G", *O	NA	5	Bipolar DC-bus compatible/uneven switch utilization, lacks isolation
Flying capacitor	interface	[46]	Traditional	8(0)	C(6) L(1)	*G", *O	NA	5	Better standalone and high frequency operation/high implementation cost

Topology	Class	Ref	Structure	Switch (Diode) Count	Passive Element (Count)	AC Mode	Active Balancing	AC Levels	Merits/Demerits
CT-MLC		[45,47]	Traditional	10/8	C(4) L(1)	*G, *O	NA	9	More levels, better fault tolerance/high switch count
HBFC	MIC with	[48]	Hybrid	14/14	C(2) L(1)	*G", *O	NA	13	Higher levels, modu- lar/unidirectional
SC-HB	two port interface	[48]	Hybrid	10/11	C(1) L(1)	*G″	NA	13	Higher levels, modular, unidirectional
G- MLC		[48]	Hybrid	8/8	C(3) L(1)	*G", *O	NA	11	Scalable, high-power density
FC-ANPC		[49]	Hybrid	12/12	C(4) L(1)	"G"	NA	11	Higher levels/unidirectional
Symmetric CHB		[62,63]	Cascaded H bridge	24/24	L(1)	*G", *O	Module level	13	Higher efficiency, low THD, faster module balancing, supports redundant operation
Asymmetric CHB		[65,67,88]	Cascaded H bridge	24/24	L(1)	*G", *O	module voltage restriction	>30	Higher levels in AC voltage/large number of cells in module. Uneven steps in the AC voltage.
RCHB		[36,38,66– 76]	Reduced Switch	16/16	C(1) L(1)	*0	Module level	13	Higher efficiency, Low THD, modulation limitation, grid-connected mode not possible
MCMLC	MLC with	[4,79,85]	Reduced Switch	13/17	L(1)	*G", *O	Module level	13	Wide operational range/voltage spikes on mode transition
PUCMLC	multi-port interface	[74]	Reduced Switch	6/6	C(1) L(1)	*G", *O	NA	7	Modular/voltage dependency of battery modules
HB/FB		[4]	Modular MLC	12/12	C(2) L(3)	*G", *O	Module level	13	Modular/fault- current blocking limitation for HB
SSC		[84]	Modular MLC	30/30	C(12) L(3)	*G", *O	Module level	13	Low volume /high switch account
CCSM		[83]	Modular MLC	12/12	C(1) L(1)	*G", *O	NA	13	Modular/voltage dependency of battery modules
АСННВ		[87]	Hybrid	18/18	C(1) L(1)	*G", *O	Module level	13	Higher efficiency, low THD, grid mode not possible
ANPC-CHB		[87]	Hybrid	18/20	C(3) L(1)	*G", *O	NA	13	Low volume/high switch account
MHFH		[87]	Hybrid	12/12	C(3) L(1)	*G", *O	NA	13	Low volume/high switch account

Table 3. Cont.

*G" (only grid injection possible), *G (gird bidirectional), *O (Off grid).

It can be seen that except for PUCMLC, all multiport MLC topologies can generate 13 levels in the AC voltage. Another exception is the asymmetric-CHB topology, which can generate more than 30 levels due to the asymmetric-voltage profile of the battery modules. However, this large number of levels in AC voltage remains uneven and the resulting THD in the fundamental AC voltage is higher than the limit permitted by the grid codes.

Table 4 highlights the comparative summary of seven multiport MLC topologies reported in [4,35,36,38,62–76,79,81,85,88,90] for different applications. The topologies are divided into three performance categories.

- (i) Appropriate with no operational limitation;
- Possible but not efficient due to high switch count, lack of modularity, or high THD in the AC voltage;
- (iii) Not suitable at all due to functional limitations.

Topology Feature	Symmetric CHB [62,63]	Asymmetric CHB [64,65,67,88]	RCHB [41]	MCMLC [35,36,38,66–76]	HB/FB MLC [4,79,81,85]	АСННВ [90]
Grid offtake			•			
Grid injection			•	•	•	•
Frequency balancing				•		
Standalone (UPS mode)				•		
Scalability						
Reliability						
Battery fault tolerance				•		
(Dual on-grid and off-grid	•			•	•	•
Footprint (No: of components)				•		
AC-fault ride-through						•
Enhancing power quality				•		•
Black start						
Voltage support				•		•
Hot swapping		•		•		
Capacity extension				•		
Different chemistry integration				•		
DC microgrid support				•		
🔵 Арр	propriate topolog	y 🦲 Possible	but not effic	ient 🥚 Not s	uitable topology	

Table 4. Comparative summary of cascaded MLC topologies, based on [4,35,36,38,62–76,79,81,85,88,90].

The subsection below highlights the merits and demerits of each topology for gridconnected and off-grid systems.

3.1. Off-Grid/Standalone Operation

Comparing all potential topologies from (i) to (vi), it is clear that the RCHB topology holds the advantage due to having the lowest number of semiconductor switches (16) and passive components (1). It can control the discharging of various battery modules independently, reach any number of cascaded HBs/voltage levels, and allow redundant operation. In addition, RHCB has good fault tolerance, as it can bypass the faulty module without affecting the performance of running modules, thus considerably improving the

reliability of the system. Another feature of RCHB and ACHHB topologies is the DC microgrid support. The intermediate multistep DC link can be used to feed an independent load or a DC microgrid.

3.2. On-Grid Operation

Unlike off-grid mode, RHCB lacks the current-control mode due to the synchronous operation of the front-end FB. As a result, it can only offtake power (uncontrolled and non-sinusoidal current) from the grid but cannot inject the power into the grid. The modular MLC topologies (MCMLC, HB, FB) can perform both offtake and injection but suffer from high passive-component counts, as they require two inductors in each arm.

ACHHB works on the same operating principle as RHCB and therefore lacks the ability to inject power into the grid.

SCHB [62,63] and ACHB [64,65,67,88] can perform both offtake and injection with no operational limitation. However, ACHB entails a large string of cells in each module (>30), which leads to large and uneven steps in the AC voltage, ultimately resulting in high THD exceeding the grid codes.

Both SCHB and ACHB topologies hold the advantage of grid-connected applications, including AC-fault ride-through, power quality, and frequency balancing, but are not the best candidates for the standalone system due to their large footprint (i.e., high switch count). However, unlike RHCB, such limitations do not impede the functional ability of SCHB/ACHB to feed a standalone load. On the other hand, the reduced-switch topologies RHCB [35,36,38,66–76] and ACHHB [90] are optimally appropriate for standalone systems but face absolute limitations when it comes to the grid injection.

In a nutshell, it can be deduced that there is no single universal topology that can be considered optimal for both grid-connected and standalone operation; however, SCHB [62,63] can perform reasonably well for dual mode of operation.

4. Modulation Strategies of Multiport MLCs

4.1. Basic Principle

The primary role of an MLC modulation strategy is to enhance the output voltage quality so that it stays close to a pure sinusoidal waveform, as explained in [36,91–96]. For three-phase systems, there can be a circulating current among the legs of an MLC, which needs to be suppressed by the modulation strategy [36]. The switching frequency of the semiconductor switches should be minimized to reduce the switching losses [97,98]. Additional requirements from the distribution system operator (DSO) mandate a grid-connected MLC to comply with the grid codes [99,100]. These requirements include injecting pure sinusoidal current at unity power factor, active and reactive power control, DC-current suppression, total harmonic distortion in current below 5% during grid-injection mode, and fault-ride-through capability for various grid faults and defects in the battery modules.

The modulation technique needs to ensure that the operating principle of active balancing is not violated while also maintaining good-quality AC voltage. This means the injection angle needs to be updated during zero transition of the AC voltage.

Figure 10 depicts the overview of the potential modulation techniques for a multiport MLC in compliance with active balancing among the battery modules.

4.2. Classification of Modulation Strategies

The modulation strategies can be broadly classified into fixed- and variable-frequency modulation, as reported in [36]. The fixed-frequency modulation can be further divided into multicarrier- and fundamental-frequency modulation. Multicarrier PWM is subdivided into level-shifted carrier (LSC)- and phase-shifted carrier (PSC)-based PWM methods, whereas in fundamental-frequency modulation, the underlying methods include selective harmonic elimination (SHE)-, nearest-level control (NLC)-, and optimized control angle (OCA)-based PWM schemes.



Figure 10. Modulation techniques for the MLC in compliance with the active-balancing role.

The variable-frequency PWM methods include hysteresis and hybrid PWM. They provide a faster dynamic response compared to fixed-frequency-based PWM techniques but are not widely used due to high switching losses.

4.2.1. Multiple-Carrier-Based PWM

In LSC, multiple-carrier signals are generated, which are shifted vertically in the modulation space, as explained in [101–104]. Typically, the carrier signals have a frequency that is 20 to 30 times the fundamental frequency of the AC output voltage. Each carrier signal corresponds to two switches of one leg.

For the FB topology shown in Figure 4, the two carrier signals are used. C_1 corresponds to S_{11} , and S_{21} whereas C_2 correspond to S_{12} and S_{22} . For the HB topology there is a single carrier signal *C* representing S_1 and S_2 . The total number of carrier signals depends on the number of cascaded FB and HB structures (12 for the CHB topology in Figures 5 and 6 for the RHCB topology in Figure 6). A single modulating reference signal coming from higher-level control (explained in the next subsection) is compared with the multiple-carrier signals to generate PWM for the switches.

Figure 11 shows various sub-techniques for LSC-based PWM schemes. A single cycle of the modulating reference signal is considered for better illustration. The x-axis shows the time in seconds, whereas the y-axis reflects different levels of carrier amplitude.



Figure 11. Various sub-techniques for LSC-based PWM scheme [105,106]: (a) equal—phase LSC PWM for FB structure, (b) opposite—phase LSC PWM for FB structure, (c) equal—phase LSC-PWM for HB structure, and (d) PSC PWM for FB structure.

In an equal-phase (EP) PWM scheme [105], the multiple-carrier signals are shifted vertically with a DC offset equal to the carrier magnitude symmetrically above and below the mean value of the modulating reference signal, as shown in Figure 11a.

In an opposite-phase (OP) PWM scheme [106], the carrier signals are shifted similar to EP-PWM but with the complementary phases. The carrier waves (C_1 , C_2) above the mean value of the modulating reference are out of phase to C_1 lying below, as shown in Figure 11b.

The DC offset corresponds to m_n and therefore directly controls the switch-in time of each module. A lower value of DC offset indicates more turn-on and less turn-off time, and vice versa.

During standalone or grid-injection mode, the battery modules with higher OCV or SOC are placed near the mean value of the modulating reference signal, whereas the modules with lower OCV or SOC occupy the top layer. Figure 10c reflects the concept reported in [70] where the carrier signals are ordered such that $C_1 > C_2 > C_3$. This means that the battery modules at the top layer (C_1) receive less switch-on time, and therefore less energy is consumed from module 1 compared to module 2 and module 3.

To reduce the switching transients, the carrier signals are reordered every half cycle or complete cycle when the modulating wave goes through the zero crossing. The reordered carrier signal becomes effective during the next cycle.

Comparing Figure 11a–c, it can be seen that the modulating reference signal (V_{mr}) for HB has a frequency two times higher than that of FB, so the carrier signals can be re-ordered twice in each cycle of the AC output voltage.

In the PSC-based PWM scheme [106], all carrier signals have the same magnitude and frequency as the modulating reference signal but are phase shifted by $\frac{180^{\circ}}{n}$ (where *n* is the number of modules) in each period of the modulating reference signal.

Unlike amplitude-based carrier-shifting techniques, each carrier signal in PSC occupies the same modulation space, as shown in Figure 11d. This allows PSC to achieve a linear modulation range for all switches and better quality of the output voltage. However, it cannot regulate the switch-in and -out duration of different battery modules due to the fixed period, which means all battery modules supply the same energy at the end of each switching period.

4.2.2. Fundamental Frequency-Based PWM

In this method, the switches are turned on once during each cycle of the fundamental AC voltage. The turning on of each switch is governed by the switching angle, which is calculated using various techniques. Prominent techniques include SHE-, NLC-, and OCA-based PWM.

In the SHE-based PWM method, the undesired odd harmonic components and distortion factor of the AC-voltage waveform is suppressed, as explained in [107,108]. This is achieved by finding the complementary firing angles α_1 and β_1 using non-linear methods.

These complementary angles result in the switch turn-on times that correspond to a pure sinusoidal waveform, as depicted in Figure 12b. The output AC-voltage waveform can be expressed in Fourier series as follows:

$$V_{AC-out} = \sum_{k=1,3,5,\dots}^{\infty} \frac{4V_B}{k\pi} \{ \cos(k\alpha_1) + \cos(k\alpha_2) + \dots + \cos(k\alpha_6) \} \sin(kwt)$$
(15)

where *k* is the number of the undesired odd harmonics and V_B is the module voltage. The magnitude of the Fourier coefficients for the fundamental voltage V_{AC-1} (k = 1) and maximum value V_{AC-1} can be expressed as follows:

$$V_{1} = \frac{4V_{B}}{\pi} \left\{ \cos\left(\alpha_{1}\right) + \cos\left(\alpha_{2}\right) + \ldots + \cos\left(\alpha_{6}\right) \right\}$$

$$V_{1-\max} = n \frac{4V_{B}}{\pi} for \alpha_{i} = 0$$

$$\left\{ 16 \right\}$$



Figure 12. Various sub-techniques for LSC PWM scheme: (**a**) opposite-phase PWM for FB structure [106], (**b**) alternate opposite-phase PWM for FB structure [106], and (**c**) synthesized nine-level voltage waveform for SHE modulation [108].

The modulation index M_{SHE} is then expressed as:

$$M_{SHE} = \frac{\pi V_{AC-1}}{n4V_B} \tag{17}$$

To cancel the odd harmonics, the magnitude of the Fourier coefficients specific to odd ones are set to zero, as shown below [108]:

$$\sum_{k=3,5,7,\dots}^{21} \{ \cos(k\alpha_1) + \cos(k\alpha_2) + \dots + \cos(k\alpha_6) \} = 0$$
(18)

From Equation (18), it can be seen that there are 10 nonlinear transcendental equations that need to be solved using iterative methods such as the Newton–Raphson method [75] to obtain the optimized value for switching angles.

Once the switching angles are obtained, the energy balance is achieved by alternately reversing the angle assignment in pairs, as explained in [107,108] As a result, $\alpha_1 = \alpha_6$, $\alpha_2 = \alpha_4$, and $\alpha_4 = \alpha_5$ after each period of the fundamental AC voltage. This allows the energy balance to be maintained among the modules. To avoid solving nonlinear equations in each AC voltage, the firing angles were calculated offline in [108] and stored in a look-up table. SHE is applicable for both FB and HB configurations.

In the nearest-level control (NLC) method, a sinusoidal reference is compared with the actual output AC voltage to select the nearest sinusoidal voltage level. NLC is computationally less intensive compared to SHE, as it does not require nonlinear equations to be solved. It also offers a better quality of output voltage and a smaller ripple in the load current [109–111].

In optimal switching angle (OCA), the harmonics are minimized by calculating the optimal value of the switching angle using nonlinear methods [36].

4.2.3. Variable-Frequency-Based PWM

k

In the underlying methods for this category, the current error signal and fixed-width reference-hysteresis band are compared to generate the switching signals for MLC switches. Refs. [112–114] explain different variable-frequency PWM strategies and assess the impact of filter elements on network harmonics.

4.3. Comparative Analysis

All the fixed-frequency-based modulation techniques explained in Section 4.2 provide adequate performance from the power-quality perspective. However, when the active-balancing functionality is added, the fundamental-frequency-based PWM methods reported in [107,108] suffer from computational complexity. For example, in the SHE method, the offline calculation of injection angles reported in [108] maintains only the energy balance among the modules. This means that each module supplies the same kWh after a definite time interval T. Therefore, it is effective only as long as all the battery modules have same capacity, initial SOC, and discharge/charge rate (C). In the case of a battery pack with different capacity modules, the energy-balancing approach forces the lower-capacity modules to operate at higher C rates, resulting in much faster discharging/charging than the rest of the battery modules.

To avoid this issue, the angle assignment needs to be updated during each switching interval based on either the SOC or OCV of each module. As explained in Section 4.2.2, when the angle-assignment time is combined with the time to solve the nonlinear equation in each AC cycle, the computational complexity increases manifold.

The multi-carrier-based PWM methods have the advantage of simple calculation but face other limitations; for example, all LSC techniques reported in [74,103,104] are inherently nonlinear and unbalanced due to the fractional share of modulation space, as shown in Figures 11 and 12. If the battery modules have fresh cells with the same initial voltage/SOC, the LSC always makes the battery modules unbalanced first before performing the active balancing.

The PSC scheme does not undergo this issue and continues the balanced operation for an extended range. However, like SHE, the PSC method can only perform the energy balance among modules and is therefore not effective if the modules are initially unbalanced or do not have the same capacity.

5. High-Level Control

5.1. Main Objective

A robust high-level controller ensures that a good tracking of the active and reactive current reference is achieved for grid-connected (current-control) mode [115,116]. In the standalone (voltage-control) mode the controller must reduce the voltage transients, dv/dt, semiconductor losses, and switching delays. A good estimation of semiconductor voltage drop and compensation of switching delays is also critical to achieve an accurate reference following on the AC side.

The controller must also maintain the system stability during active balancing, module bypassing, or hot swapping. Additionally, it must perform a smooth transition from grid-connected mode to standalone mode.

The output of a high-level controller is a sinusoidal modulating-reference signal that drives the modulation block.

In grid-connected mode the high-level controller can also perform power-quality conditioning, which includes current-harmonics suppression and reactive-power compensation for nonlinear loads to enhance the grid power quality.

5.2. Classification of Higher-Level Control Strategies

Figure 13 provides a classification of various high-level control strategies that can be used for multiport MLCs.

5.2.1. Synchronous Reference-Frame (SRF)-Based Control

In the sub types of SRF-based control methods, voltage-oriented control (VOC) is quite common for grid-connected MLCs [117].

In VOC, all the synchronous quantities are first converted to the orthogonal α - β stationary reference frame using the Park transformation and then to DC components in a rotating *dq* reference frame using the Clark transformation [118,119]. Figure 14 shows the block diagram of VOC control reported in [120] for a single-phase MLC.



Figure 13. High-level control techniques for MLCs.



Figure 14. High-level VOC controller for CHB MLC [118].

A single-phase phase-locked loop (PLL) is used to synchronize the rotating dq frame with the angle (θ) of the fundamental grid voltage. For cascaded battery modules a single current-control loop is used.

Two proportional-integral (PI) controllers are used to control the i_d (d-axis component representing active power) and i_q (q-axis component representing reactive power). The output of the PI controllers is added to the feedforward signals for cross-coupling elimination. The output corresponds to V_{ref-d} and V_{ref-q} , which are the dq components of the reference voltage. After the reverse Park transformation, the two orthogonal reference output voltages $V_{ref-\alpha}$ and $V_{ref-\beta}$ are obtained. $V_{ref-\alpha}$ is used as a modulating-input signal for the modulation stage.

The direct current controller (DCC) is reported in [121] for m-level PV-fed MLCs but can be applied to a BESS-fed MLC topology as well. It provides an accurate current control in grid-connected mode and allows for robust performance under system-fault conditions. Figure 15 depicts the controller reported in [122]. DCC uses a sinc filter as reported in [122,123] along with the multiple second-order generalized integrators (multi-SOGI) to filter the measured grid voltage and AC-side current. The output of the current

 $I_{\alpha\beta}-ref \qquad Current \qquad V_{module}$ $I_{\alpha\beta}-ref \qquad controller \qquad \downarrow$ $I_{\alpha\beta}-ref \qquad controller \qquad \downarrow$ $Balan \\ cing \qquad \downarrow$ $V_{grid} \qquad Sinc \qquad Multi \qquad Abc \qquad V_{\alpha}-ref \qquad Or Multi \qquad Gating signals$ $V_{grid} \qquad Triangle$

controller can be used with the module-balancing algorithms, which select the switching vector so as to minimize the distortion in the output voltage.

Figure 15. Schematic view of high-level direct-current controller [121].

5.2.2. Predictive Current Control (PCC)

PCC uses the discretized MLC-model behavior in the previous time instant to predict the output at next time instant.

It makes multiple control-index predictions in each sampling interval and chooses the one that leads to the minimum deviation between the desired output and the predicted value. Generally, the control target is to track a pure sinusoidal AC reference in phase with the grid voltage.

PCC is reported in [124–127], where forward Euler discretization is first used to obtain the discrete-time dynamic model of the MLC. The tracking error of predicted-current values is governed by cost function C, which can be expressed as follows:

$$C_{k+1} = i_{k+1}^* - i_{k+1} \left| i_{k+1}^* - i_{k+1} \right|_2^2$$
(19)

selection

where i_{k+1}^* is the desired value at k + 1 time instant, and $|.|_2^2$ represents the quadratic eucledian norm.

In PCC, the sampling time is kept small to hold the approximation of $i_{k+1}^* = i_k$ true. The resulting actual current i_{k+1} at k + 1 time instant is the same as i_{k+1}^* .

To avoid common-mode voltage generation for three-phase implementation, PCC incorporates the uncertainty or mismatch between the actual and sampled filter parameters. To reduce the calculation time for PCC, a fast-optimization algorithm based on sphere decoding is reported in [126,127].

5.2.3. Non-Synchronous Reference-Frame (PR)-Based Control

In this method, the frame transformation does not take place. The actual and reference current values are compared in a stationary reference frame and the difference is fed to a proportional resonant (PR) controller [128,129]. The PR controller is tunned at the fundamental frequency of the grid. In the case of harmonic suppression, multiple PR controllers can be used. Each controller is tuned at a frequency that is an odd multiple of the fundamental grid frequency. The output of the PR controller is the modulating reference signal, which drives the modulator block.

PR-based controllers face higher tuning issues than PI-based controllers but offer high speed, as they do not need frame transformation.

5.2.4. Nonlinear Control

Nonlinear controllers feature smaller amplitude variations and offer quick response but face design challenges when load parameters are dynamic and change over time, such as off-grid systems. The sub-techniques in this category include sliding-mode control (SMC), hysteresis controllers, and deadbeat controllers.

SMC is reported in [130,131] for cascaded three-phase dual-inverter systems and cascaded H-bridge, respectively.

In SMC, first a sliding surface is defined using nonequality constraints such as commonmode voltage reduction and cancellation of higher-order harmonics. Once the constraints are defined, the control variable, which is usually the MLC output current, is squeezed onto the sliding surface s(t), as expressed below:

$$\mathbf{s}(\mathbf{t}) = c_1 \mathbf{e}(\mathbf{t}) + \frac{d}{dt} \mathbf{e}(t) \tag{20}$$

where $e(t) = I_{ref} - i(t)$ is the error signal obtained by comparing the reference and actual MLC output current. Figure 16 reflects the schematic overview of the SMC with a low-pass filter for a CHB MLC, as reported in [132]. The low-pass filter is used to prevent overshoot and slipping around the sliding surface. To define the stability condition for the sliding surface, the Lyapunov stability standard is used in [132] and is expressed below:

$$\frac{d}{dt}e(t) = -\in sign(s) - ks - \omega \frac{R}{L}i(t) + i\ddot{(t)}$$
(21)

where $\in > 0$, sign(s) is the signum function, and k is the inequality constraint, which is defined as follows:

$$k > \left[\omega \frac{R}{L}i(t) + i(t)\right]$$
(22)



Figure 16. Schematic representation of the sliding-mode control scheme with a low-pass filter, reported in [130,131].

Another important nonlinear controller is a hysteresis controller. This is used in applications where the requirements include fast transient response, unconditional stability, and wide setpoint-tracking bandwidth [133].

In the hysteresis control method, the controlled parameter i(t) (MLC output current) is made to follow the reference current by applying a suitable voltage at the MLC output. Each voltage level corresponds to a single hysteresis limit. If i(t) exceeds the initial hysteresis limit, the next (higher or lower) voltage level is selected to force i(t) again within the specified limit. If i(t) remains out of bounds for all the voltages levels in one direction, the process repeats with reverse polarity and stops only when a particular voltage level is selected that reverses the direction of i(t).

One challenge for hysteresis controllers is to select the correct voltage level that keeps the MLC output current within the desired band. Quite often, a single-band hysteresis controller faces limitations in controlling the current. To solve this issue, a multiband hysteresis-modulation technique is reported in [134,135] that uses multiple symmetrical hysteresis bands to control the output current.

Figure 17 shows the multiband hysteresis controller reported in [134]. In this technique, the inner hysteresis band carries out switching between adjacent levels, whereas the outer band carries out additional switching-level changes.



Figure 17. Multiband hysteresis controller reported in [134].

The last nonlinear controller reviewed for MLCs is the deadbeat controller. Deadbeat controllers are reported in [136] as an alternate choice for synchronous reference-framebased controllers. They are built in stationary alpha-beta reference frames to minimize the complexity and computation burden for transformation to and from *dq* reference frames. Figure 18 reflects the deadbeat controller that is reported in [136].



Figure 18. Deadbeat current controller reported in [136].

The deadbeat action forces the MLC current i(t) to follow the reference current in each sampling interval. The deadbeat transfer function for i_{alpha} and i_{beta} can be expressed as follows:

$$\frac{i_{\alpha\beta}}{i_{\alpha\beta-ref}} = \frac{1}{1+\mathrm{Ts}} = \frac{1}{1+(\mathrm{T}+\mathrm{C})\mathrm{s}}$$
(23)

where C is a constant and T is the time delay of a single sample. The constant C is set as the same value as T in order to cancel the delay time between reference and actual current, as reported in [137].

5.3. Comparative Summary

Due to simplicity, response speed, and ease of implementation, the hysteresis control method outperforms all other methods. It exhibits robustness and a reduction in system order compared to the rest of the methods. However, it faces a serious drawback due to variable-switching frequency, as it causes non-uniform electrical and thermal stress across all the switches and entails a high cost for the heat sink.

The non-synchronous controller features fast reference tracking, but a capacity mismatch among the battery modules may lead to overmodulation and degraded power quality for standalone applications.

It is important to note that the switching frequency of an MLC is significantly lower than standard two-level VSI, which means a relatively higher sampling time is available to update the reference signal. This means that the steady-state performance and transient stability are more critical parameters than the speed of response.

It is explained in the previous subsection that both VOC- and PCC-based methods [47,119] achieve zero steady-state error and ensure system stability for a dual mode of operation. Therefore, VOC-based high-level control is considered for the simulation study in this review paper.

6. Simulation Study

In this section, a simulation study is conducted using (i) the topology based on the explanation in Section 3, (ii) the modulation methods based on the explanation in Section 4.3, and (iii) higher-level control based on the explanation in Section 5.3.

It can be seen from Table 4 that the CHB MLC topology holds the advantage over other topologies in many features. Therefore, 13-level grid-connected CHB configuration was selected for the study, as depicted in Figure 19. Based on the explanation in Section 5.3, a higher-level closed-loop controller using the synchronous reference frame was developed in a MATLAB environment.



Figure 19. Schematic representation of single-phase CHB-MLC with voltage and current measurements.

Multicarrier-based modulation methods show improved performance compared to other PWM schemes, as explained in Section 4.3, but there are multiple subtypes, and each type has different features.

The simulation study compares these subtypes, including (i) EP-, (ii) OP-, (iii) AOP-, and (iv) PS-based PWM methods.

The performance indicators include balancing speed, voltage THD, current THD, and maximum voltage difference among the unbalanced modules. The THD of MLC voltage and current were measured using fast Fourier-transform (FFT) analysis. Table 5 shows the different system parameters considered for the simulation study.

System Parameters						
Number of modules	6					
Number of cells per module	16					
Nominal module voltage	65 V					
Cell type	LFP					
Rated power	5 kW					
Nominal cell capacity	5 Ah					
Number of AC voltage levels L_{AC}	13					
Carrier-signal frequency f_C	1100 Hz					
Fundamental frequency f_o	50 Hz					
Modulating ration	{-1,1}					
Filter inductance <i>L</i> _{AC}	200 uH					
Filter resistance <i>R</i> _{AC}	0.1 mΩ					
Full-bridge switch type	MOSFET					
Switch ON resistance	10 mΩ					
Total number of switches	24					
Grid voltage (rms)	230 V					
Proportional-gain PI controller	0.4					
Integral-gain PI controller	50					
Control sampling time	50 μs					

Table 5. System parameters considered for the simulation study.

To evaluate high-level control and active balancing for different PWM methods, three tests were conducted, as follows:

- I. Monitor the AC-side parameters using a short-duration test at low and rated current. The measurements include current-tracking error and voltage/current THD).
- II. Monitor the impact of active balancing on the DC voltage of all six battery modules using a long-duration charge–discharge test with balanced modules with the same initial OCV.
- III. Monitor the impact of active balancing on unbalanced battery modules with different initial OCV and SOC.

The difference among module voltages was measured at the beginning and end of each cycle. Further details of the balancing algorithm, including the OCV/SOC calculation and a reference-modulating signal update, are beyond the scope of this paper.

Figure 20 reflects various results for the EP-PWM method. Figure 20a shows the MLC output voltage, whereas Figure 20b reflects the MLC output current and grid voltage on the same graph. The grid voltage was scaled down (by a factor of 15) to improve the visualization.



Figure 20. Simulation results for equal-phase LSC modulation: (**a**) MLC output voltage, (**b**) injected current and grid voltage, and (**c**) twelve carrier signals and modulating—reference signal from the high—level controller.

From t = 0 s to t = 0.2 s, the MLC remained in the grid-injection mode, where it injected a constant active power of 6 kW into the grid at unity power factor. This corresponds to the reference current $I_{d_ref} = 30$ A and $I_{q_ref} = 0$ A for the high-level controller. The modulating reference (output of high-level controller) and multiple career signals are depicted in Figure 19c.

It can be seen from Figure 20b that the MLC injected current was in phase with the grid voltage during the grid-injection mode. At t = 0.2 s the MLC went into offtake mode and the MLC offtook 6 kW from the grid. The reference current changed from +30 A to -30 A. The actual current followed the reference, as is clear from Figure 20b, where the MLC current was 180° out of phase with the grid voltage. The THD of the MLC voltage was 10.6% throughout t the test, whereas the output current had a THD of 1.6% in grid-injection mode and 2% in offtake mode.

In the second test, the initial OCVs of the battery modules 1 - 6 were set to the same value of 65 V, whereas the initial SOC of all the modules was kept at 80%. Figure 21a reflects the battery-module voltage profile for a charge–discharge cycle. It can be seen that the module voltages remained balanced, with a small difference (ΔV) of less than 0.01 V throughout the operation.



Figure 21. Battery-module voltage profile during a charge–discharge-cycle test for EP-PWM with (**a**) no initial imbalance in the module voltage and (**b**) a maximum initial imbalance of 5.5 V.

In the third test, the initial OCVs and SOC of battery modules 1 - 6 were set as 70.3 V, 69.2 V, 68.1 V, 67.1 V, 65.9 V and 64.8 V, and 75%, 74%, 73.5%, 72.5%, 71% and 70.5%, respectively. The test ran for a duration of 100 s. The maximum initial OCV difference was measured as $V_{mod1} - V_{mod6} = 5.5$ V. The discharging cycle continued for a duration of 50 s. The EP-PWM method gradually reduced the difference to 2.4 V at the end of the discharging period. The difference was eventually reduced to zero after 24 s of the charging cycle, as shown in Figure 21b.

Figure 22 reflects the different test results of the OP-PWM method. In the test with unbalanced modules, the same initial SOC and OCV profiles as those used for the EP-PWM method were considered.



Figure 22. (a) Battery-module voltage profile during a charge-discharge-cycle test for OP-PWM method with initial imbalance of 5.5 V and (b) battery-module current profile for two modules with maximum and minimum OCV.

Figure 22a depicts the module voltage profile for the test (iii) with unbalanced modules. The initial imbalance was measured as 5.5 V. The maximum difference at the end of the 50 s discharging cycle was measured as 2.5 V, whereas the at the end of the 24 s charging cycle the difference was reduced to 0 V.

To illustrate the impact of active balancing on the current of battery modules, the current profiles of modules 1 and 6 are depicted in Figure 22b. These two modules were selected because module 1 had the highest initial OCV, whereas module 6 had the lowest initial OCV.

It can be seen that battery module 1 had a higher average current in the discharging mode as it conducted throughout the switching period, whereas module 6 only conducted the switching current fractionally (less than 5 ms) during the switching cycle. This means that module 1 underwent rapid discharging compared to module 6, which is clear from the voltage profile of module 6 in Figure 21a.

Figure 23 reflects the battery-module voltage and current profile of the AOP-PWM method for the test (iii) with unbalanced modules. The initial imbalance was measured as 5.5 V. The maximum difference at the end of the 50 s discharging cycle was measured as 2.5 V, whereas at the end of the 24 s charging cycle the difference was not reduced to 0 and was measured as 0.3 V.

Table 6 provides an overview of different parameters for four multicarrier-based PWM methods. Using fast Fourier transform (FFT), the MLC voltage and current THD were calculated at rated and low (less than 20%) power. At rated power, the AOP-PWM method exhibited the highest voltage THD but the lowest current THD, whereas the EP-PWM method showed the lowest voltage THD but the highest current THD.



Figure 23. (**a**) Battery-module voltage profile during a charge-discharge-cycle test for AOP-PWM method with initial imbalance of 5.5 V, and (**b**) battery-module current profile for two modules with maximum and minimum OCV.

Table 6.	Comparative summary	y for different	amplitude-based	modulation strategies

Modulation-Technique Feature	EP PWM [105,106]	OP PWM [105,106]	AOP PWM [105,106]	<i>PSC</i> [105,106]
MLC voltage THD at low power	12.6%	12.72%	16.4%	10.6%
MLC current THD at low power	5.6%	5.66%	5.3%	2.6%
MLC voltage THD at rated power	10.6%	12.26%	19.6%	10.6%
MLC current THD at rated power	2.6%	2.3%	1.7%	2.6%
Balancing potential	Yes	Yes	Yes	Partially
ΔV for discharge cycle	2.5 V	2.5 V	2.5 V	NA
ΔV for charge cycle	0 V	0 V	0.3 V	NA

Compared to the EP and OP based PWM methods, AOP–PWM was less effective at reducing the imbalance factor to zero for the reference charge–discharge profile.

7. Conclusions

MLCs are increasingly applied as BESS interfaces due to their improved voltage qualities over conventional two-level voltage-source inverters. In this paper, several MLC topologies are explored for grid-connected and off-grid applications. A critical overview of traditional and recently proposed multiport topologies is presented to offer a better understanding of MLC operation and limitation in grid-connected and off-grid modes. The principle of active balancing is explained to show the implementation of the configurations in grid-connected and off-grid modes. The key findings can be summarized as follows:

- (1) The state-of-the-art reduced-switch MLC topologies reported in the literature [35,36,38,66–76] are good candidates for only standalone applications and cannot inject power in the grid-connected mode. Two such topologies are reduced-switch cascaded half-bridge [73] and modified cascaded configurations [35,38].
- (2) There is no universal topology that can be attributed as the best candidate for a dual (grid-connected and standalone) mode of operation. However, except for a relatively higher footprint, the symmetric cascaded H-bridge can be an optimal topology for a dual mode of operation.
- (3) Many modulation strategies that are used for multiport MLCs maintain energy balance only among the battery modules. Therefore, these PWM strategies lack effectiveness when combined with the role of active balancing among unbalanced modules with different capacities/chemistries. These include all phase-shifted PWM methods and fundamental-frequency PWM methods with offline-angle calculations [108].
- (4) Amplitude-based carrier PWM methods are inherently unbalanced, and the carrier signals need to be readjusted every few cycles even if the modules have the same initial OCV/SOC.
- (5) At low power, the phase-shifted PWM strategy offers the lowest voltage and current THD (10% and 2.6%, respectively) compared to level-shifted carrier-based techniques. At rated power, AOP-PWM offers the lowest current THD of 1.7%.
- (6) The balancing potential of the AOP-PWM strategy is 33% lower than the EP and OP-PWM methods.

To summarize, the diversity in cascaded MLC modulation, control, and switching strategies can allow for further improvement in the AC-side transient response and reduction in the losses. For future studies, a loss analysis can be conducted for state-of-the-art modulation strategies. In addition, during the idle state, either for grid-connected or standalone mode, the self-balancing feature of CHB-MLC can be further explored. To allow for faster balancing, it is critical to allow for a simultaneous charging/discharging feature, which means that high-capacity modules can discharge into low-capacity modules. A hybrid modulation, control, and switching strategy can be developed to allow such a feature to expedite the integration of battery modules with different capacities in stationary applications.

Author Contributions: Conceptualization, A.M.R.; methodology, A.M.R.; software, A.M.R.; validation, A.M.R. and M.A.-M.; formal analysis, A.M.R. and M.A.-M.; investigation, A.M.R.; resources, A.M.R.; data curation, A.M.R.; writing—original draft preparation, A.M.R.; writing—review and editing, A.M.R., M.A.-M., T.G. and O.H.; visualization, A.M.R., M.A.-M., T.G. and O.H.; supervision, A.M.R., M.A.-M., T.G. and O.H.; project administration, A.M.R., M.A.-M., T.G. and O.H.; funding acquisition, T.G. and O.H. All authors have read and agreed to the published version of the manuscript.

Funding: This This paper is supported by the OPEVA project that has received funding within the Key Digital Technologies Joint Undertaking (KDT JU) from the European Union's Horizon Europe Programme and the National Authorities (France, Belgium, Czechia, Italy, Portugal, Turkey, Switzerland), under grant agreement 101097267. Views and opinions expressed are however those of the author(s) only and do not necessarily reflect those of the European Union or KDT JU. Neither the European Union nor the granting authority can be held responsible for them.

Data Availability Statement: Not applicable.

Acknowledgments: Authors acknowledge Flanders Make for the support to this research group.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Bakeer, A.; Chub, A.; Shen, Y.; Sangwongwanich, A. Reliability Analysis of Battery Energy Storage System for Various Stationary Applications. *J. Energy Storage* 2022, *50*, 104217. [CrossRef]
- Faessler, B. Stationary, Second Use Battery Energy Storage Systems and Their Applications: A Research Review. *Energies* 2021, 14, 2335. [CrossRef]
- Singh, R.; Nakka, J. Design, Simulation and Performance Analysis of Fuel Cell Based Energy System with Cascaded H-Bridge Multilevel Inverter. In Proceedings of the 2016 IEEE 1st International Conference on Power Electronics, Intelligent Control and Energy Systems (ICPEICES), Delhi, India, 4–6 July 2016; pp. 1–6.
- Wang, Y.; Aksoz, A.; Geury, T.; Ozturk, S.B.; Kivanc, O.C.; Hegazy, O. A Review of Modular Multilevel Converters for Stationary Applications. *Appl. Sci.* 2020, 10, 7719. [CrossRef]
- Wang, L.; Wu, Q.H.; Tang, W. Novel Cascaded Switched-Diode Multilevel Inverter for Renewable Energy Integration. *IEEE Trans.* Energy Convers. 2017, 32, 1574–1582. [CrossRef]
- Schram, W.L.; Lampropoulos, I.; van Sark, W.G. Photovoltaic Systems Coupled with Batteries That Are Optimally Sized for Household Self-Consumption: Assessment of Peak Shaving Potential. *Appl. Energy* 2018, 223, 69–81. [CrossRef]
- Luthander, R.; Widén, J.; Munkhammar, J.; Lingfors, D. Self-Consumption Enhancement and Peak Shaving of Residential Photovoltaics Using Storage and Curtailment. *Energy* 2016, *112*, 221–231. [CrossRef]
- 8. Jansson, S. Evaluation of KPIs and Battery Usage of Li-Ion BESS for FCR Application; Uppsala Universitet: Uppsala, Sweden, 2019.
- 9. Rancilio, G.; Rossi, A.; Di Profio, C.; Alborghetti, M.; Galliani, A.; Merlo, M. Grid-Scale BESS for Ancillary Services Provision: SoC Restoration Strategies. *Appl. Sci.* 2020, *10*, 4121. [CrossRef]
- 10. Alaperä, I.; Hakala, T.; Honkapuro, S.; Pylvänäinen, J.; Kaipia, T.; Manner, P.; Kulla, T. Battery System as a Service for a Distribution System Operator; AIM: Canberra, Australia, 2019.
- 11. Choi, J.W.; Aurbach, D. Promise and Reality of Post-Lithium-Ion Batteries with High Energy Densities. *Nat. Rev. Mater.* **2016**, 1, 16013. [CrossRef]
- 12. Choi, D.; Shamim, N.; Crawford, A.; Huang, Q.; Vartanian, C.K.; Viswanathan, V.V.; Paiss, M.D.; Alam, M.J.E.; Reed, D.M.; Sprenkle, V.L. Li-Ion Battery Technology for Grid Application. *J. Power Sources* **2021**, *511*, 230419. [CrossRef]
- Hidalgo-León, R.; Siguenza, D.; Sanchez, C.; León, J.; Jácome-Ruiz, P.; Wu, J.; Ortiz, D. A Survey of Battery Energy Storage System (BESS), Applications and Environmental Impacts in Power Systems. In Proceedings of the 2017 IEEE Second Ecuador Technical Chapters Meeting (ETCM), Salinas, Ecuador, 16–20 October 2017; pp. 1–6.
- 14. Arias, N.B.; López, J.C.; Hashemi, S.; Franco, J.F.; Rider, M.J. Multi-Objective Sizing of Battery Energy Storage Systems for Stackable Grid Applications. *IEEE Trans. Smart Grid* 2020, 12, 2708–2721. [CrossRef]
- 15. Zhang, Y.; Gevorgian, V.; Wang, C.; Lei, X.; Chou, E.; Yang, R.; Li, Q.; Jiang, L. Grid-Level Application of Electrical Energy Storage: Example Use Cases in the United States and China. *IEEE Power Energy Mag.* **2017**, *15*, 51–58. [CrossRef]
- 16. +Pavić, I.; Luburić, Z.; Pandžić, H.; Capuder, T.; Andročec, I. Defining and Evaluating Use Cases for Batter*y Energy Storage Investments: Case Study in Croatia. *Energies* **2019**, *12*, 376.
- Lawder, M.T.; Suthar, B.; Northrop, P.W.C.; De, S.; Hoff, C.M.; Leitermann, O.; Crow, M.L.; Santhanagopalan, S.; Subramanian, V.R. Battery Energy Storage System (BESS) and Battery Management System (BMS) for Grid-Scale Applications. *Proc. IEEE* 2014, 102, 1014–1030. [CrossRef]
- Alhejaj, S.M.; Gonzalez-Longatt, F.M. Investigation on Grid-Scale BESS Providing Inertial Response Support. In Proceedings of the 2016 IEEE International Conference on Power System Technology (POWERCON), Wollongong, NSW, Australia, 28 September–1 October 2016; pp. 1–6.
- Fernández-Muñoz, D.; Pérez-Díaz, J.I.; Guisández, I.; Chazarra, M.; Fernández-Espina, Á. Fast Frequency Control Ancillary Services: An International Review. *Renew. Sustain. Energy Rev.* 2020, 120, 109662. [CrossRef]
- 20. Sandelic, M.; Stroe, D.-I.; Iov, F. Battery Storage-Based Frequency Containment Reserves in Large Wind Penetrated Scenarios: A Practical Approach to Sizing. *Energies* 2018, *11*, 3065. [CrossRef]
- Kim, J.-H.; Choi, K.-Y.; Kim, R.-Y. A Low Frequency Input Current Reduction Scheme of a Two-Stage Single-Phase Inverter with DC-DC Boost Converter. In Proceedings of the 2014 IEEE Applied Power Electronics Conference and Exposition-APEC 2014, Fort Worth, TX, USA, 16–20 March 2014; pp. 2351–2358.
- 22. Lo, K.-Y.; Chen, Y.-M.; Chang, Y.-R. Bidirectional Single-Stage Grid-Connected Inverter for a Battery Energy Storage System. *IEEE Trans. Ind. Electron.* **2016**, *64*, 4581–4590. [CrossRef]
- Kim, H.; Parkhideh, B.; Bongers, T.D.; Gao, H. Reconfigurable Solar Converter: A Single-Stage Power Conversion PV-Battery System. *IEEE Trans. Power Electron.* 2012, 28, 3788–3797. [CrossRef]
- Ponnaluri, S.; Linhofer, G.O.; Steinke, J.K.; Steimer, P.K. Comparison of Single and Two Stage Topologies for Interface of BESS or Fuel Cell System Using the ABB Standard Power Electronics Building Blocks. In Proceedings of the 2005 European Conference on Power Electronics and Applications, Dresden, Germany, 11–14 September 2005; p. 9.
- 25. Kıvrak, S.; Özer, T.; Oğuz, Y.; Kelek, M.M. Novel Active and Passive Balancing Method-Based Battery Management System Design and Implementation. *J. Power Electron.* **2021**, *21*, 1855–1865. [CrossRef]
- Ziegler, A.; Oeser, D.; Arndt, B.; Ackva, A. Comparison of Active and Passive Balancing by a Long Term Test Including a Post-Mortem Analysis of All Single Battery Cells. In Proceedings of the 2018 International IEEE Conference and Workshop in Óbuda on Electrical and Power Engineering (CANDO-EPE), Budapest, Hungary, 20–21 November 2018; pp. 15–20.

- Vitols, K. Redesign of Passive Balancing Battery Management System to Active Balancing with Integrated Charger Converter. In Proceedings of the 2014 14th Biennial Baltic Electronic Conference (BEC), Tallinn, Estonia, 6–8 October 2014; pp. 241–244.
- Aizpuru, I.; Iraola, U.; Mari, J.; Goikoetxea, A. Comparative Study and Evaluation of Passive Balancing against Single Switch Active Balancing Systems for Energy Storage Systems. In Proceedings of the PCIM Europe 2016; International Exhibition and Conference for Power Electronics, Intelligent Motion, Renewable Energy and Energy Management, Nuremberg, Germany, 10–12 May 2016; VDE: Frankfurt, Germany, 2016; pp. 1–9.
- Palma, L.; Enjeti, P. An Inverter Output Filter to Mitigate DV/Dt Effects in PWM Drive System. In Proceedings of the APEC. Seventeenth Annual IEEE Applied Power Electronics Conference and Exposition (Cat. No. 02CH37335), Dallas, TX, USA, 10–14 March 2002; Volume 1, pp. 550–556.
- 30. Colak, I.; Kabalci, E.; Bayindir, R. Review of Multilevel Voltage Source Inverter Topologies and Control Schemes. *Energy Convers. Manag.* 2011, 52, 1114–1128. [CrossRef]
- Abu-Rub, H.; Holtz, J.; Rodriguez, J.; Baoming, G. Medium-Voltage Multilevel Converters—State of the Art, Challenges, and Requirements in Industrial Applications. *IEEE Trans. Ind. Electron.* 2010, 57, 2581–2596. [CrossRef]
- 32. Amankwah, E.K.; Watson, A.J.; Clare, J.C. Operation of a Hybrid Modular Multilevel Converter during Grid Voltage Unbalance. *IET Gener. Transm. Distrib.* **2016**, *10*, 3102–3110. [CrossRef]
- Behrouzian, E.; Bongiorno, M.; De La Parra, H.Z. An Overview of Multilevel Converter Topologies for Grid Connected Applications. In Proceedings of the 2013 15th European Conference on Power Electronics and Applications (EPE), Lille, France, 2–6 September 2013; pp. 1–10.
- Bughneda, A.; Salem, M.; Richelli, A.; Ishak, D.; Alatai, S. Review of Multilevel Inverters for PV Energy System Applications. Energies 2021, 14, 1585. [CrossRef]
- Liu, M.; Cao, X.; Cao, C.; Wang, P.; Wang, C.; Pei, J.; Lei, H.; Jiang, X.; Li, R.; Li, J. A Review of Power Conversion Systems and Design Schemes of High-Capacity Battery Energy Storage Systems. *IEEE Access* 2022, *10*, 52030–52042. [CrossRef]
- Sinha, A.; Jana, K.C.; Das, M.K. An Inclusive Review on Different Multi-Level Inverter Topologies, Their Modulation and Control Strategies for a Grid Connected Photo-Voltaic System. *Sol. Energy* 2018, 170, 633–657. [CrossRef]
- 37. R-anjan, R.K.; Tiwari, P.N. Detailed Review on Multilevel Inverters Driving Renewable Energy Systems with PWM Modulation Technique. *Res. J. Eng. Technol. Med. Sci.* 2022, *5*, 250–256.
- Mohamad, A.S.; Mariun, N.; Sulaiman, N.; Radzi, M.A.M. A New Cascaded Multilevel Inverter Topology with Minimum Number of Conducting Switches. In Proceedings of the 2014 IEEE Innovative Smart Grid Technologies-Asia (ISGT Asia), Kuala Lumpur, Malaysia, 20–23 May 2014; pp. 164–169.
- Eyer, J.; Corey, G. Energy Storage for the Electricity Grid: Benefits and Market Potential Assessment Guide. Sandia Natl. Lab. 2010, 20, 5.
- Hesse, H.C.; Schimpe, M.; Kucevic, D.; Jossen, A. Lithium-Ion Battery Storage for the Grid—A Review of Stationary Battery Storage System Design Tailored for Applications in Modern Power Grids. *Energies* 2017, 10, 2107. [CrossRef]
- 41. Taallah, A.; Mekhilef, S. Active Neutral Point Clamped Converter for Equal Loss Distribution. *IET Power Electron.* 2014, 7, 1859–1867. [CrossRef]
- Rodriguez, J.; Bernet, S.; Steimer, P.K.; Lizama, I.E. A Survey on Neutral-Point-Clamped Inverters. *IEEE Trans. Ind. Electron.* 2009, 57, 2219–2230. [CrossRef]
- Burguete, E.; López, J.; Zabaleta, M. A New Five-Level Active Neutral-Point-Clamped Converter with Reduced Overvoltages. IEEE Trans. Ind. Electron. 2016, 63, 7175–7183. [CrossRef]
- Feng, C.; Liang, J.; Agelidis, V.G. Modified Phase-Shifted PWM Control for Flying Capacitor Multilevel Converters. *IEEE Trans.* Power Electron. 2007, 22, 178–185. [CrossRef]
- 45. Jana, J.; Saha, H.; Bhattacharya, K. Das A Review of Inverter Topologies for Single-Phase Grid-Connected Photovoltaic Systems. *Renew. Sustain. Energy Rev.* 2017, 72, 1256–1270. [CrossRef]
- Adam, G.P.; Alajmi, B.; Ahmed, K.H.; Finney, S.J.; Williams, B.W. New Flying Capacitor Multilevel Converter. In Proceedings of the 2011 IEEE International Symposium on Industrial Electronics, Gdansk, Poland, 27–30 June 2011; pp. 335–339.
- Jana, K.C.; Biswas, S.; KarChaudhary, S. Dual Reference Phase Shifted PWM Technique for a N-Level Inverter Based Grid Connected Solar Photovoltaic System. *IET Renew. Power Gener.* 2016, 10, 928–935.
- Babaei, E.; Gowgani, S.S. Hybrid Multilevel Inverter Using Switched Capacitor Units. *IEEE Trans. Ind. Electron.* 2013, 61, 4614–4621. [CrossRef]
- Lezana, P.; Aceitón, R. Hybrid Multicell Converter: Topology and Modulation. *IEEE Trans. Ind. Electron.* 2010, 58, 3938–3945. [CrossRef]
- 50. Mehta, P.; Kunapara, A.; Karelia, N. Improvement in Switching Strategy Used for Even Loss Distribution in ANPC Multilevel Inverter. *Procedia Technol.* 2015, *21*, 386–392. [CrossRef]
- 51. Gholizad, A.; Farsadi, M. A Novel State-of-Charge Balancing Method Using Improved Staircase Modulation of Multilevel Inverters. *IEEE Trans. Ind. Electron.* **2016**, *63*, 6107–6114. [CrossRef]
- Young, C.-M.; Chu, N.-Y.; Chen, L.-R.; Hsiao, Y.-C.; Li, C.-Z. A Single-Phase Multilevel Inverter with Battery Balancing. *IEEE Trans. Ind. Electron.* 2012, 60, 1972–1978. [CrossRef]
- 53. Gautam, S.P.; Kumar, L.; Gupta, S. Single-phase Multilevel Inverter Topologies with Self-voltage Balancing Capabilities. *IET Power Electron.* **2018**, *11*, 844–855. [CrossRef]

- 54. Lee, S.-W.; Lee, K.-M.; Choi, Y.-G.; Kang, B. Modularized Design of Active Charge Equalizer for Li-Ion Battery Pack. *IEEE Trans. Ind. Electron.* **2018**, *65*, 8697–8706. [CrossRef]
- 55. Nelson, P.A.; Santini, D.J.; Barnes, J. Factors Determining the Manufacturing Costs of Lithium-Ion Batteries for PHEVs. *World Electr. Veh. J.* **2009**, *3*, 457–468. [CrossRef]
- 56. Yu, J.; Zhang, G.; Peng, M.; Song, D.; Liu, M. Power-Matching Based SOC Balancing Method for Cascaded H-Bridge Multilevel Inverter. *CPSS Trans. Power Electron. Appl.* **2020**, *5*, 352–363. [CrossRef]
- 57. Peng, F.Z. A Generalized Multilevel Inverter Topology with Self Voltage Balancing. *IEEE Trans. Ind. Appl.* **2001**, *37*, 611–618. [CrossRef]
- 58. Maharjan, L.; Yamagishi, T.; Akagi, H. Active-Power Control of Individual Converter Cells for a Battery Energy Storage System Based on a Multilevel Cascade PWM Converter. *IEEE Trans. Power Electron.* **2010**, *27*, 1099–1107. [CrossRef]
- 59. Du, Z.; Tolbert, L.M.; Ozpineci, B.; Chiasson, J.N. Fundamental Frequency Switching Strategies of a Seven-Level Hybrid Cascaded H-Bridge Multilevel Inverter. *IEEE Trans. Power Electron.* **2009**, *24*, 25–33. [CrossRef]
- Wei, S.; Wu, B.; Li, F.; Sun, X. Control Method for Cascaded H-Bridge Multilevel Inverter with Faulty Power Cells. In Proceedings of the Eighteenth Annual IEEE Applied Power Electronics Conference and Exposition, 2003—APEC'03, Miami Beach, FL, USA, 9–13 February 2003; Volume 1, pp. 261–267.
- Villanueva, E.; Correa, P.; Rodríguez, J.; Pacas, M. Control of a Single-Phase Cascaded H-Bridge Multilevel Inverter for Grid-Connected Photovoltaic Systems. *IEEE Trans. Ind. Electron.* 2009, *56*, 4399–4406. [CrossRef]
- 62. Sarkar, I.; Fernandes, B.G. A Hybrid Symmetric Cascaded H-Bridge Multilevel Converter Topology. *IEEE J. Emerg. Sel. Top. Power Electron.* **2021**. [CrossRef]
- Raj, N.; Anand, A.; Riyas, A.; Jagadanand, G.; George, S. A Novel Open-Transistor Fault Detection Method in Symmetric Cascaded H-Bridge Multilevel Inverter. In Proceedings of the 2016 IEEE International Conference on Power Electronics, Drives and Energy Systems (PEDES), Trivandrum, India, 14–17 December 2016; pp. 1–6. [CrossRef]
- 64. Mortezaei, A.; Simões, M.G.; Bubshait, A.S.; Busarello, T.D.C.; Marafão, F.P.; Al-Durra, A. Multifunctional Control Strategy for Asymmetrical Cascaded H-Bridge Inverter in Microgrid Applications. *IEEE Trans. Ind. Appl.* **2016**, *53*, 1538–1551. [CrossRef]
- 65. Jiang, W.; Ren, K.; Xue, S.; Yang, C.; Xu, Z. Research on the Asymmetrical Multilevel Hybrid Energy Storage System Based on Hybrid Carrier Modulation. *IEEE Trans. Ind. Electron.* **2020**, *68*, 1241–1251. [CrossRef]
- 66. Wu, F.; Li, X.; Feng, F.; Gooi, H.B. Modified Cascaded Multilevel Grid-Connected Inverter to Enhance European Efficiency and Several Extended Topologies. *IEEE Trans. Industr Inform.* **2015**, *11*, 1358–1365. [CrossRef]
- 67. Babaei, E.; Hosseini, S.H.; Gharehpetian, G.B.; Haque, M.T.; Sabahi, M. Reduction of Dc Voltage Sources and Switches in Asymmetrical Multilevel Converters Using a Novel Topology. *Electr. Power Syst. Res.* **2007**, *77*, 1073–1085. [CrossRef]
- Chang, F.; Zheng, Z.; Li, Y. PWM Strategy of a Novel Cascaded Multi-Level Converter for Battery Management. In Proceedings of the 2014 17th International Conference on Electrical Machines and Systems (ICEMS), Hangzhou, China, 22–25 October 2014; pp. 3208–3212. [CrossRef]
- 69. Sonti, V.; Dhara, S.; Kukade, P.; Jain, S.; Agarwal, V. Analysis for the Minimization of Leakage and Common Mode Currents in Cascaded Half-Bridge PV Fed Multilevel Inverter. *IEEE J. Emerg. Sel. Top. Power Electron.* **2018**, *7*, 2443–2452. [CrossRef]
- 70. Mei, J.; Xiao, B.; Shen, K.; Tolbert, L.M.; Zheng, J.Y. Modular Multilevel Inverter with New Modulation Method and Its Application to Photovoltaic Grid-Connected Generator. *IEEE Trans. Power Electron.* **2013**, *28*, 5063–5073. [CrossRef]
- Xiao, B.; Hang, L.; Mei, J.; Riley, C.; Tolbert, L.M.; Ozpineci, B. Modular Cascaded H-Bridge Multilevel PV Inverter with Distributed MPPT for Grid-Connected Applications. *IEEE Trans. Ind. Appl.* 2014, *51*, 1722–1731. [CrossRef]
- Zheng, Z.; Wang, K.; Peng, L.; Li, Y.; Xu, L. A Hybrid Cascaded Multi-Level Converter for Power Storage System. In Proceedings of the 2013 15th European Conference on Power Electronics and Applications (EPE), Lille, France, 2–6 September 2013; pp. 1–10. [CrossRef]
- Zheng, Z.; Wang, K.; Xu, L.; Li, Y. A Hybrid Cascaded Multilevel Converter for Battery Energy Management Applied in Electric Vehicles. *IEEE Trans. Power Electron.* 2013, 29, 3537–3546. [CrossRef]
- Konstantinou, G.S.; Agelidis, V.G. Performance Evaluation of Half-Bridge Cascaded Multilevel Converters Operated with Multicarrier Sinusoidal PWM Techniques. In Proceedings of the 2009 4th IEEE Conference on Industrial Electronics and Applications, Xi'an, China, 25–27 May 2009; pp. 3399–3404.
- 75. Ounejjar, Y.; Al-Haddad, K.; Gregoire, L.-A. Packed U Cells Multilevel Converter Topology: Theoretical Study and Experimental Validation. *IEEE Trans. Ind. Electron.* **2010**, *58*, 1294–1306. [CrossRef]
- 76. Metri, J.I.; Vahedi, H.; Kanaan, H.Y.; Al-Haddad, K. Real-Time Implementation of Model-Predictive Control on Seven-Level Packed U-Cell Inverter. *IEEE Trans. Ind. Electron.* **2016**, *63*, 4180–4186. [CrossRef]
- Gao, Z.; Lu, Q. A Hybrid Cascaded Multilevel Converter Based on Three-Level Cells for Battery Energy Management Applied in Electric Vehicles. *IEEE Trans. Power Electron.* 2018, 34, 7326–7349. [CrossRef]
- 78. Nami, A.; Liang, J.; Dijkhuizen, F.; Demetriades, G.D. Modular Multilevel Converters for HVDC Applications: Review on Converter Cells and Functionalities. *IEEE Trans. Power Electron.* **2014**, *30*, 18–36. [CrossRef]
- 79. Debnath, S.; Qin, J.; Bahrani, B.; Saeedifard, M.; Barbosa, P. Operation, Control, and Applications of the Modular Multilevel Converter: A Review. *IEEE Trans. Power Electron.* **2014**, *30*, 37–53. [CrossRef]
- 80. Lee, S.S.; Sidorov, M.; Lim, C.S.; Idris, N.R.N.; Heng, Y.E. Hybrid Cascaded Multilevel Inverter (HCMLI) with Improved Symmetrical 4-Level Submodule. *IEEE Trans. Power Electron.* **2017**, *33*, 932–935. [CrossRef]

- 81. Adam, G.P.; Davidson, I.E. Robust and Generic Control of Full-Bridge Modular Multilevel Converter High-Voltage DC Transmission Systems. *IEEE Trans. Power Deliv.* 2015, *30*, 2468–2476. [CrossRef]
- 82. Bhesaniya, M.M.; Shukla, A. Norton Equivalent Modeling of Current Source MMC and Its Use for Dynamic Studies of Back-to-Back Converter System. *IEEE Trans. Power Deliv.* **2016**, *32*, 1935–1945. [CrossRef]
- Nami, A.; Wang, L.; Dijkhuizen, F.; Shukla, A. Five Level Cross Connected Cell for Cascaded Converters. In Proceedings of the 2013 15th European Conference on Power Electronics and Applications (EPE), Lille, France, 2–6 September 2013; pp. 1–9. [CrossRef]
- 84. Elserougi, A.A.; Massoud, A.M.; Ahmed, S. A Switched-Capacitor Submodule for Modular Multilevel HVDC Converters with DC-Fault Blocking Capability and a Reduced Number of Sensors. *IEEE Trans. Power Deliv.* **2015**, *31*, 313–322. [CrossRef]
- 85. Zhang, J.; Jiang, C.; Han, Y. Modular Multilevel Converter Composite Submodule Topology and Control. *J. Eng.* 2019, 2019, 2643–2648. [CrossRef]
- Sandeep, N.; Yaragatti, U.R. Operation and Control of a Nine-Level Modified ANPC Inverter Topology with Reduced Part Count for Grid-Connected Applications. *IEEE Trans. Ind. Electron.* 2017, 65, 4810–4818. [CrossRef]
- Roy, J.; Chamarthi, P.; Agarwal, V. A Hybrid 9-Level Inverter with Minimum Number of Switches for Single Phase Grid Connected Solar PV System. In Proceedings of the 2016 IEEE 43rd Photovoltaic Specialists Conference (PVSC), Portland, OR, USA, 5–10 June 2016; pp. 3252–3257. [CrossRef]
- 88. Chattopadhyay, S.K.; Chakraborty, C. A New Asymmetric Multilevel Inverter Topology Suitable for Solar PV Applications with Varying Irradiance. *IEEE Trans. Sustain. Energy* 2017, *8*, 1496–1506. [CrossRef]
- Sandeep, N.; Yaragatti, U.R. Design and Implementation of a Sensorless Multilevel Inverter with Reduced Part Count. *IEEE Trans. Power Electron.* 2017, 32, 6677–6683. [CrossRef]
- 90. Gupta, K.K.; Jain, S. A Novel Multilevel Inverter Based on Switched DC Sources. *IEEE Trans. Ind. Electron.* 2013, *61*, 3269–3278. [CrossRef]
- 91. Quraan, M.; Tricoli, P.; D'Arco, S.; Piegari, L. Efficiency Assessment of Modular Multilevel Converters for Battery Electric Vehicles. *IEEE Trans. Power Electron.* **2016**, *32*, 2041–2051. [CrossRef]
- 92. Mhiesan, H.; Wei, Y.; Siwakoti, Y.P.; Mantooth, H.A. A Fault-Tolerant Hybrid Cascaded H-Bridge Multilevel Inverter. *IEEE Trans. Power Electron.* **2020**, *35*, 12702–12715. [CrossRef]
- Gao, F.; Zhang, L.; Zhou, Q.; Chen, M.; Xu, T.; Hu, S. State-of-Charge Balancing Control Strategy of Battery Energy Storage System Based on Modular Multilevel Converter. In Proceedings of the 2014 IEEE Energy Conversion Congress and Exposition (ECCE), Pittsburgh, PA, USA, 14–18 September 2014; pp. 2567–2574.
- 94. De Simone, D.; Tricoli, P.; D'Arco, S.; Piegari, L. Windowed PWM: A Configurable Modulation Scheme for Modular Multilevel Converter-Based Traction Drives. *IEEE Trans. Power Electron.* **2020**, *35*, 9727–9736. [CrossRef]
- 95. Iyer, K.V.; Mohan, N. Modulation and Commutation of a Single Stage Isolated Asymmetrical Multilevel Converter for the Integration of Renewables and Battery Energy Storage System in Ships. *IEEE Trans. Transp. Electrif.* **2016**, *2*, 580–596. [CrossRef]
- 96. Leon, J.I.; Portillo, R.; Vazquez, S.; Padilla, J.J.; Franquelo, L.G.; Carrasco, J.M. Simple Unified Approach to Develop a Time-Domain Modulation Strategy for Single-Phase Multilevel Converters. *IEEE Trans. Ind. Electron.* **2008**, *55*, 3239–3248. [CrossRef]
- Shen, K.; Zhao, D.; Mei, J.; Tolbert, L.M.; Wang, J.; Ban, M.; Ji, Y.; Cai, X. Elimination of Harmonics in a Modular Multilevel Converter Using Particle Swarm Optimization-Based Staircase Modulation Strategy. *IEEE Trans. Ind. Electron.* 2014, 61, 5311–5322. [CrossRef]
- Guo, X.; Wang, X.; Wang, C.; Lu, Z.; Hua, C.; Blaabjerg, F. Improved Modulation Strategy for Singe-Phase Cascaded H-Bridge Multilevel Inverter. *IEEE Trans. Power Electron.* 2021, 37, 2470–2474. [CrossRef]
- 99. Ma, K.; Muñoz-Aguilar, R.S.; Rodriguez, P.; Blaabjerg, F. Thermal and Efficiency Analysis of Five-Level Multilevel-Clamped Multilevel Converter Considering Grid Codes. *IEEE Trans. Ind. Appl.* **2013**, *50*, 415–423.
- Babaie, M.; Abarzadeh, M.; Al-Haddad, K. An Improved High-Resolution Wide Bandwidth ANPC Converter Using VLMM and Lyapunov Stability Theory for Grid-Connected Applications. In Proceedings of the 2022 IEEE Energy Conversion Congress and Exposition (ECCE), Detroit, MI, USA, 9–13 October 2022; pp. 1–6.
- Sarkar, I.; Fernandes, B.G. Modified Hybrid Multi-Carrier PWM Technique for Cascaded H-Bridge Multilevel Inverter. In Proceedings of the IECON 2014-40th Annual Conference of the IEEE Industrial Electronics Society, Dallas, TX, USA, 29 October–1 November 2014; pp. 4318–4324.
- 102. Aghdam, M.G.H.; Fathi, S.H.; Gharehpetian, G.B. Analysis of Multi-Carrier PWM Methods for Asymmetric Multi-Level Inverter. In Proceedings of the 2008 3rd IEEE Conference on Industrial Electronics and Applications, Singapore, 3–5 June 2008; pp. 2057–2062.
- 103. McGrath, B.P.; Holmes, D.G. Multicarrier PWM Strategies for Multilevel Inverters. *IEEE Trans. Ind. Electron.* 2002, 49, 858–867. [CrossRef]
- Kaliannan, T.; Albert, J.R.; Begam, D.M.; Madhumathi, P. Power Quality Improvement in Modular Multilevel Inverter Using for Different Multicarrier PWM. *Eur. J. Electr. Eng. Comput. Sci.* 2021, 5, 19–27. [CrossRef]
- Ramaprabha, R.; Malathy, S.; Krishnan, N.R. A 17-Level Asymmetric Multilevel Inverter with Fundamental Frequency PWM Schemes. In Proceedings of the 2019 2nd International Conference on Power and Embedded Drive Control (ICPEDC), Chennai, India, 21–23 August 2019; pp. 510–515.
- Radan, A.; Shahirinia, A.H.; Falahi, M. Evaluation of Carrier-Based PWM Methods for Multi-Level Inverters. In Proceedings of the 2007 IEEE International Symposium on Industrial Electronics, Vigo, Spain, 4–7 June 2007; pp. 389–394.

- 107. Dahidah, M.S.A.; Konstantinou, G.; Agelidis, V.G. A Review of Multilevel Selective Harmonic Elimination PWM: Formulations, Solving Algorithms, Implementation and Applications. *IEEE Trans. Power Electron.* **2014**, *30*, 4091–4106. [CrossRef]
- Monem, M.A.; Hegazy, O.; Omar, N.; Van Mierlo, J.; Van Den Bossche, P.; Mulder, G.; Mantels, B. Influence of Selective Harmonic Elimination Technique of Multi-Level DC/DC Converter on Second-Life Battery Performances. In Proceedings of the 2013 International Conference on Renewable Energy Research and Applications (ICRERA), Madrid, Spain, 20–23 October 2013; pp. 1107–1111.
- Thakre, M.P.; Jadhav, T.K.; Patil, S.S.; Butale, V.R. Modular Multilevel Converter with Simplified Nearest Level Control (NLC) Strategy for Voltage Balancing Perspective. In Proceedings of the 2021 Innovations in Energy Management and Renewable Resources (52042), Kolkata, India, 5–7 February 2021; pp. 1–8.
- Nguyen, M.H.; Kwak, S. Nearest-Level Control Method with Improved Output Quality for Modular Multilevel Converters. IEEE Access 2020, 8, 110237–110250. [CrossRef]
- 111. Meshram, P.M.; Borghate, V.B. A Simplified Nearest Level Control (NLC) Voltage Balancing Method for Modular Multilevel Converter (MMC). *IEEE Trans. Power Electron.* **2014**, *30*, 450–462. [CrossRef]
- Tsai, C.-T.; Chen, S.-H. PV Power-Generation System with a Phase-Shift PWM Technique for High Step-up Voltage Applications. *Int. J. Photoenergy* 2012, 2012, 838231. [CrossRef]
- 113. Lee, J.-D.; Park, D.-H.; Kim, R.-Y. Novel Variable Switching Frequency PWM Strategy for a SiC-MOSFET-Based Electric Vehicle Inverter to Increase Battery Usage Time. *IEEE Access* 2022, *10*, 21929–21940. [CrossRef]
- 114. Rangarajan, S.S.; Collins, E.R.; Fox, J.C. Comparative Impact Assessment of Filter Elements Associated with PWM and Hysteresis Controlled PV on Network Harmonic Resonance in Distribution Systems. In Proceedings of the 2017 IEEE 6th International Conference on Renewable Energy Research and Applications (ICRERA), San Diego, CA, USA, 5–8 November 2017; pp. 80–85.
- 115. Jia, J.; Yang, G.; Nielsen, A.H. A Review on Grid-Connected Converter Control for Short-Circuit Power Provision under Grid Unbalanced Faults. *IEEE Trans. Power Deliv.* 2017, 33, 649–661. [CrossRef]
- 116. Taul, M.G.; Wang, X.; Davari, P.; Blaabjerg, F. An Overview of Assessment Methods for Synchronization Stability of Grid-Connected Converters under Severe Symmetrical Grid Faults. *IEEE Trans. Power Electron.* **2019**, *34*, 9655–9670. [CrossRef]
- El-Hosainy, A.; Hamed, H.A.; Azazi, H.Z.; El-Kholy, E.E. A Review of Multilevel Inverter Topologies, Control Techniques, and Applications. In Proceedings of the 2017 Nineteenth International Middle East Power Systems Conference (MEPCON), Cairo, Egypt, 19–21 December 2017; pp. 1265–1275.
- Bhattacharya, S.; Divan, D. Synchronous Frame Based Controller Implementation for a Hybrid Series Active Filter System. In Proceedings of the IAS '95—Conference Record of the 1995 IEEE Industry Applications Conference Thirtieth IAS Annual Meeting, Orlando, FL, USA, 8–12 October 1995; Volume 3, pp. 2531–2540.
- 119. Kesler, M.; Ozdemir, E. Synchronous-Reference-Frame-Based Control Method for UPQC Under Unbalanced and Distorted Load Conditions. *IEEE Trans. Ind. Electron.* 2011, *58*, 3967–3975. [CrossRef]
- Chatzinikolaou, E.; Rogers, D.J. Cell SoC Balancing Using a Cascaded Full-Bridge Multilevel Converter in Battery Energy Storage Systems. *IEEE Trans. Ind. Electron.* 2016, 63, 5394–5402. [CrossRef]
- Ambrozic, V.; Fiser, R.; Nedeljkovic, D. Direct Current Control-a New Current Regulation Principle. *IEEE Trans. Power Electron.* 2003, 18, 495–503. [CrossRef]
- Korn, A.J.; Winkelnkemper, M.; Steimer, P. Low Output Frequency Operation of the Modular Multi-Level Converter. In Proceedings of the 2010 IEEE Energy Conversion Congress and Exposition, Atlanta, GA, USA, 12–16 September 2010; pp. 3993–3997.
- 123. Anuchin, A.; Lashkevich, M.; Shpak, D.; Aliamkin, D.; Zharkov, A.; Briz, F. Current Control of AC Drives Using Shunt Current Sensors and Delta-Sigma Modulation. In Proceedings of the IECON 2018—44th Annual Conference of the IEEE Industrial Electronics Society, Washington, DC, USA, 21–23 October 2018; pp. 445–449.
- Geyer, T.; Quevedo, D.E. Multistep Direct Model Predictive Control for Power Electronics—Part 1: Algorithm. In Proceedings of the 2013 IEEE Energy Conversion Congress and Exposition, Denver, CO, USA, 15–19 September 2013; pp. 1154–1161.
- Geyer, T.; Quevedo, D.E. Multistep Direct Model Predictive Control for Power Electronics—Part 2: Analysis. In Proceedings of the 2013 IEEE Energy Conversion Congress and Exposition, Denver, CO, USA, 15–19 September 2013; pp. 1162–1169.
- Cortés, P.; Wilson, A.; Kouro, S.; Rodriguez, J.; Abu-Rub, H. Model Predictive Control of Multilevel Cascaded H-Bridge Inverters. IEEE Trans. Ind. Electron. 2010, 57, 2691–2699. [CrossRef]
- 127. Aguilera, R.P.; Baidya, R.; Acuna, P.; Vazquez, S.; Mouton, T.; Agelidis, V.G. Model Predictive Control of Cascaded H-Bridge Inverters Based on a Fast-Optimization Algorithm. In Proceedings of the IECON 2015—41st Annual Conference of the IEEE Industrial Electronics Society, Yokohama, Japan, 9–12 November 2015; pp. 4003–4008.
- 128. Zhou, Y.; Jiang, D.; Guo, J.; Hu, P.; Liang, Y. Analysis and Control of Modular Multilevel Converters Under Unbalanced Conditions. *IEEE Trans. Power Deliv.* 2013, 28, 1986–1995. [CrossRef]
- 129. Li, J.; Konstantinou, G.; Wickramasinghe, H.R.; Pou, J. Operation and Control Methods of Modular Multilevel Converters in Unbalanced AC Grids: A Review. *IEEE J. Emerg. Sel. Top. Power Electron.* **2019**, *7*, 1258–1271. [CrossRef]
- Ishfaq, M.; Uddin, W.; Zeb, K.; Islam, S.U.; Hussain, S.; Khan, I.; Kim, H.J. Active and Reactive Power Control of Modular Multilevel Converter Using Sliding Mode Controller. In Proceedings of the 2019 2nd International Conference on Computing, Mathematics and Engineering Technologies (iCoMET), Sukkur, Pakistan, 30–31 January 2019; pp. 1–5.
- Yang, Q.; Saeedifard, M.; Perez, M.A. Sliding Mode Control of the Modular Multilevel Converter. *IEEE Trans. Ind. Electron.* 2019, 66, 887–897. [CrossRef]

- 132. Nguyen, V.Q.; Le, T.L.; Nguyen, M.T. Sliding Mode Control for Cascaded Multilevel Inverters. In Proceedings of the 2019 International Conference on System Science and Engineering (ICSSE), Dong Hoi, Vietnam, 20–21 July 2019; pp. 495–500.
- Corzine, K.A. A Hysteresis Current-Regulated Control for Multi-Level Drives. *IEEE Trans. Energy Convers.* 2000, 15, 169–175. [CrossRef]
- 134. Shukla, A.; Ghosh, A.; Joshi, A. Hysteresis Modulation of Multilevel Inverters. *IEEE Trans. Power Electron.* **2010**, *26*, 1396–1409. [CrossRef]
- 135. Marchesoni, M. High-Performance Current Control Techniques for Application to Multilevel High-Power Voltage Source Inverters. *IEEE Trans. Power Electron.* **1992**, *7*, 189–204. [CrossRef]
- Atia, Y.; Ahmed, M.; Eliwa, A.; Salem, M. Deadbeat Current Controller Design for Multilevel Grid Connected Inverter. Energy Procedia 2016, 100, 237–242. [CrossRef]
- 137. Martin, D.; Santi, E. Autotuning of Digital Deadbeat Current Controllers for Grid-Tie Inverters Using Wide Bandwidth Impedance Identification. *IEEE Trans. Ind. Appl.* **2013**, *50*, 441–451. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.